

Contents lists available at ScienceDirect

Progress in Oceanography



journal homepage: www.elsevier.com/locate/pocean

Projecting climate change impacts from physics to fisheries: A view from three California Current fisheries

James A. Smith^{a,b,*}, Mercedes Pozo Buil^{b,c}, Barbara Muhling^{a,b}, Desiree Tommasi^{a,b}, Stephanie Brodie^{b,c}, Timothy H. Frawley^{b,c}, Jerome Fiechter^d, Stefan Koenigstein^{b,c}, Amber Himes-Cornell^e, Michael A. Alexander^f, Steven J. Bograd^{b,c}, Nathalí Cordero Quirós^{b,c}, Larry B. Crowder^g, Enrique Curchitser^h, Stephanie J. Greenⁱ, Natasha A. Hardy^{g,i}, Alan C. Haynie^j, Elliott L. Hazen^{b,c}, Kirstin Holsman^j, Gwendal Le Fol^{a,b}, Nerea Lezama-Ochoa^{b,c}, Ryan R. Rykaczewski^k, Charles A. Stock¹, Stephen Stohs^a, Jonathan Sweeney^k, Heather Welch^{b,c}, Michael G. Jacox^{c,f}

^a NOAA Southwest Fisheries Science Center, La Jolla, CA, USA

^b Institute of Marine Sciences, University of California Santa Cruz, Santa Cruz, CA, USA

^c NOAA Southwest Fisheries Science Center, Monterey, CA, USA

^d Ocean Sciences Department, University of California Santa Cruz, Santa Cruz, CA, USA

^e Fisheries and Aquaculture Division, Food and Agriculture Organization of the United Nations, Rome, Italy

^f NOAA Physical Sciences Laboratory, Boulder, CO, USA

⁸ Hopkins Marine Station, Stanford University, Pacific Grove, CA, USA

^h Department of Environmental Sciences, Rutgers University, New Brunswick, NJ, USA

¹ Department of Biological Sciences, University of Alberta, Edmonton, AB, Canada

^j NOAA Alaska Fisheries Science Center, Seattle, WA, USA

^k NOAA Pacific Islands Fisheries Science Center, Honolulu, HI, USA

¹NOAA Geophysical Fluid Dynamics Laboratory, Princeton, NJ, USA

A R T I C L E I N F O

Keywords: Climate prediction Biological oceanography Fishery management Habitat selection Spatial planning ROMS

ABSTRACT

Motivated by a need for climate-informed living marine resource management, increased emphasis has been placed on regional end-to-end modeling frameworks designed to project climate impacts on marine ecosystems and evaluate the efficacy of potential management strategies under changing conditions. The 'Future Seas' project was initiated with a focus on three fisheries (Pacific sardine, swordfish, and albacore tuna) in the California Current System (CCS). This work leverages a suite of climate, ocean, ecosystem, and economic models to project physical, ecological, and socio-economic change, evaluate management strategies, and quantify uncertainty in model projections. Here we describe the components of the modeling framework, considerations underlying choices made in model development, engagement with stakeholders, and key physical, ecological, and socio-economic results to date, including projections to 2100. Our broad aims are to (i) synthesize a large body of climate and fisheries research that has been conducted, and continues, under the Future Seas umbrella, and (ii) provide insight and recommendations to those pursuing similar efforts for other applications and in other regions. In general, our results indicate that all three species will likely shift their distributions (predominantly poleward) in the future, which impacts accessibility to fishing fleets, spatial management, and quota allocation. For similar integrative climate-to-fisheries projections, we recommend attention is given to: recognizing potential biases arising from differences between the climate products used for ecological model fitting and those used for model projection; how sources of projection uncertainty are prioritized, incorporated, and communicated; and quantitatively linking scenarios - especially socio-economic scenarios - with climate and ecological projections.

* Corresponding author at: Institute of Marine Sciences, University of California Santa Cruz, Santa Cruz, CA, USA. *E-mail address:* jsmith35@ucsc.edu (J.A. Smith).

https://doi.org/10.1016/j.pocean.2023.102973

Received 20 July 2022; Received in revised form 1 December 2022; Accepted 3 January 2023 Available online 20 January 2023 0079-6611/© 2023 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

1. Introduction

Climate change is impacting marine systems and the communities and economies that rely on them globally (IPCC, 2022). Impacts include changes to ocean physics and biogeochemistry, species distributions and abundance, fisheries catch, and food web structure and function (Doney et al., 2012; Poloczanska et al., 2013; Pecl et al., 2017; Lotze et al., 2019; du Pontavice et al., 2020; Cooley et al., 2022). Altered ecosystem states can translate to changes in global fisheries production and uncertain economic futures (Blenckner et al., 2015; Free et al., 2019; Lam et al., 2016; Ding et al., 2017; Cheung, 2018). The United Nations declared 2021-2030 the 'Ocean Decade' in recognition of the unprecedented challenges facing sustainable ocean development, and a core goal of the Decade is predicting ocean conditions and their impact on human wellbeing and livelihoods (Ryabinin et al., 2019). Climate model projections are a critical foundation for responding to and planning for a changing marine environment, and investigations of the social-ecological system informed by regional ocean projections can help us meet our global, regional, and national obligations towards ecosystem and social resilience and sustainability (Busch et al., 2016; Hollowed et al., 2020). While components of these social-ecological systems are often examined individually, coordinated efforts that link processes from physics to fisheries can be highly effective when applied strategically to address key knowledge gaps (Fulton, 2010).

The 'Future Seas' project (https://future-seas.com) focuses on enhancing climate-ready management by linking high-resolution ocean models of the California Current System (CCS; Pozo Buil et al., 2021) with ecological and social models. Future Seas was built 'from the bottom up' by producing a base of modeled physical and biochemical CCS conditions, to which subsequent ecological and economic models are connected (Fig. 1). The end goal is to produce coupled climateecological-social models to evaluate management strategies resilient to future change in the CCS. Of the seven common types of coupled socialecological models applied for ecosystem-based management identified by Kasperski et al. (2021), the Future Seas project has to date used four: Management Strategy Evaluation (MSE), fisher behavior, social vulnerability, and end-to-end. This first phase of the project has focused on quantifying rates of change in the physical and biogeochemical environment and the potential impacts of this change on three U.S. West Coast fisheries, as well as exploring strategic climate-ready management options to increase their sustainability. The ocean projections are produced by a high-resolution ocean model coupled to a biogeochemical model, and are available for use in any type of coupled modeling, making our recommendations and lessons learned about climate-ready management in the CCS directly applicable to other efforts in this region.

2. Aims and organization of this article

Our overarching aims are to summarize the structure and results of Future Seas as well as to provide guidance for others pursuing similar efforts. To that end, we present the modeling framework and justifications for its design, main findings to date, next steps for projecting physical-ecological-social change in the CCS, and general recommendations that are broadly applicable to similar projects.

Sections 3 and 4 convey key considerations when initiating this type of project and how they were applied under Future Seas. Section 3 outlines the setting of Future Seas, which provides the context and motivation for its design, and Section 4 outlines the modelling methodology. These sections provide an example of how one can build a modeling infrastructure based on the physical, ecological, and socioeconomic context of specific fisheries. Sections 5 and 6 are tailored to readers who are interested specifically in the three CCS fisheries addressed under phase one of the Future Seas project. Section 5 summarizes physical, ecological, socioeconomic, and management results to date, and Section 6 discusses future plans. Section 7 gives recommendations that are generally applicable for those pursuing similar projects tailored to other regions, species, and management questions. Key definitions and acronyms are defined in Table 1.



Fig. 1. Schematic of the first phase of the Future Seas project. High resolution ocean models are used to generate retrospective ocean hindcasts and re-analyses, and to dynamically downscale global climate projections to provide the foundation for a range of modeling approaches aimed at quantifying and projecting aspects of three species and their U.S. West Coast fisheries: Pacific sardine (the purse seine fishery), swordfish (the drift gillnet fishery), and North Pacific albacore (the surface hook-and-line fishery). Silhouettes of species indicate relevant models and outputs. For output extents, blue indicates output forced by climate projections, yellow by ocean reanalyses, and gray indicates no climate models were used. Coupled social-ecological outputs are identified according to type: MSE, end-to-end, fisher behavior, and social vulnerability. Key definitions and acronyms are defined in Table 1. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

Table 1

Definition of key terms and acronyms in the context of this paper. The relationship between the reanalysis and projection results in terms of model fitting and prediction is especially important and is discussed in Section 7.3. Some definitions are based partly on those in IPCC (2021a).

Term	Definition
Hindcast Reanalysis	A historical simulation, forced by boundary conditions that are constrained by observations (e.g., accurate winds, air temperature, etc.). Provides a realistic representation of historical conditions and a baseline against which future change is measured. Similar to a hindcast, provides a realistic representation
	of historical conditions. In addition to the realistic boundary conditions used in the hindcast, a reanalysis also assimilates available observations (e.g., from satellites or in situ platforms) to further improve fidelity of the model to nature.
Forecast	Prediction of what conditions will be some time in the future (typically days to several years). Forecast success depends on skillfully representing the ocean state at the forecast's initialization, and simulating the ocean's evolution over the forecast window. Data assimilation is thus often used to improve forecast initialization and subsequent forecast skill. Unlike climate projections, forecasts are intended to represent the actual ocean conditions in specific years.
Prediction	In climate and physics, prediction is synonymous with forecast. In ecology, prediction is output from a fitted model(s) calculated using new input data, so projections, forecasts, and hindcasts are all types of prediction.
Projection	A long-term (multidecadal) simulation of the ocean's response to future emissions of greenhouse gases and aerosol concentrations. Ocean projections are intended to assess the evolution of the multi-decadal mean and variability of ocean conditions (i.e., the "ocean climate"), not to predict the conditions during a specific year. They are dependent on, for example, future emission scenarios and technological and socio-economic developments, which may or may not be realized.
Downscaling	Using statistical and/or dynamical methods (e.g., high- resolution regional ocean models) to translate coarse scale information (e.g., from a global climate model) to finer resolution information. Downscaling typically also includes a bias correction step to adjust for biases in the global model (e.g. Section 4.1).
Management strategy evaluation	A type of closed-loop simulation which includes variability in ecological and/or management processes (Puntet al., 2016). The simulation has one or more operating models used to predict the ecological system (e. g. the abundance of fish). This is coupled to an assessment model which simulates the human dimension (e.g. fishing) and management strategies (e.g. catch quotas), which then feeds back into the operating model (e.g. through an updated fishing mortality). It is termed 'closed-loop' because feedback from the assessment model influences the state of the operating model at each time step.

Acronyms: CCS: California Current system; CCLME: California Current large marine ecosystem; COG: center of gravity; CPS: coastal pelagic species; DGN: drift gillnet fishery; ESM: earth system model; GCM: general circulation model (also global climate model); HMS: highly migratory species; IATTC: Inter-American Tropical Tuna Commission; IBM: individual based model; MSE: management strategy evaluation; NEMUCSC: an adapted version of the North Pacific ecosystem model for understanding regional oceanography (NEMURO); PFMC: Pacific Fisheries Management Council; PNW: Pacific Northwest; RCP: representative concentration pathway; ROMS: regional ocean modeling system; ROMS-GFDL: downscaled projection forced by the NOAA Geophysical Fluid Dynamics Laboratory ESM2M model; ROMS-IPSL: downscaled projection forced by the Institut Pierre Simon Laplace CM5A-MR model; ROMS-HAD: downscaled projection forced by the Met Office Hadley Center HadGEM2-ES model; SDM: species distribution model; WCPFC: Western and Central Pacific Fishery Commission.

3. The physical, ecological, and socio-economic setting of future seas

In a project, like Future Seas, which describes and models parts of a complex system, it is important to understand the broader setting. This setting should motivate and constrain design of the modeling and analysis framework, and provide broader context for the results. Here, three fisheries in the CCS provide an example of how physical, ecological, and fishery information – including stakeholder input – can be considered in the design of end-to-end climate-fisheries projects.

3.1. Key aspects of the environmental setting

The CCS is a dynamic and productive eastern boundary upwelling system with strong linkages among large-scale climate, local ocean conditions, and living marine resource dynamics. It stretches from southern Canada to northern Mexico, encompassing the entire west coast of the contiguous U.S. (Fig. 2). The CCS exhibits strong patterns of high- and low-frequency variation in ocean climate and productivity (Baumgartner et al., 1992; Checkley and Barth, 2009; Koslow et al., 2014), and recently a series of intense warming events has impacted numerous species and fisheries (Cavole et al., 2016; McCabe et al., 2016; Wells et al., 2017; Santora et al., 2020; Harvey et al., 2020; Fisher et al., 2021; Weber et al., 2021). Projections of conditions in the CCS indicate that climate change will continue to impact the physical, biogeochemical, and ecological realms (e.g. Rykaczewski and Dunne, 2010; Marshall et al., 2017; Morley et al., 2018; Howard et al., 2020; Pozo Buil et al., 2021; IPCC, 2021b), including changes in the timing, intensity and spatial heterogeneity of upwelling (Rykaczewski et al., 2015; Brady et al., 2017; Xiu et al., 2018). These projected changes, and their attendant uncertainty (e.g., Brady et al., 2017), necessitate management strategies that are robust and responsive to a variable and changing climate. The development of these strategies is aided by projections of potential future physical and biogeochemical conditions in the CCS, which can be regionally tailored (e.g. through dynamical downscaling) to represent key mesoscale ocean features and processes (Hollowed et al., 2009; Drenkard et al., 2021; Pozo Buil et al., 2021).

3.2. Key aspects of the ecological setting

The CCS hosts a complex food web (Horne et al., 2010; Koehn et al., 2016), and a diversity of forage species supports a rich assemblage of predators while acting as a buffer to productivity cycles of some forage species (Madigan et al., 2012; Kaplan et al., 2017). There are strong links between the environment and lower-trophic level fish abundance and productivity, but the large seasonal, annual, and decadal scale variation in conditions makes identifying mechanisms challenging (King et al., 2011). As a result, there is mixed support for synchrony (i.e. common environmental sensitivity) in fish assemblage structure and commercial catches in the region (Thompson et al., 2019; Siple et al., 2020; Ong et al., 2021). In addition, the interplay between fishing and climate fluctuations may increase the rate of change in forage populations (Essington et al., 2015). For higher trophic level predators, the CCS has been described as both attractive and retentive due to the nutrient-rich upwelled waters, drawing species that migrate to the region to forage or breed (Block et al., 2011). Climate change is predicted to bring new changes in prey availability and predator distribution throughout the CCS (Hazen et al., 2013; Cheung et al., 2015; Morley et al., 2018).

The first phase of the Future Seas project focuses on three species and their main commercial fisheries in the U.S. CCS (Fig. 1): Pacific sardine (*Sardinops sagax*), swordfish (*Xiphias gladius*), and albacore (*Thunnus alalunga*). Each species is economically valuable in the CCS, and selecting them as case studies allows us to explore climate change impacts given diverse ecological drivers, fishery characteristics, and management priorities.

Sardine is an iconic species of the CCS, characterized by highly



Fig. 2. Map of the ROMS domain (black box, with color representing SST from the ROMS reanalysis for a sample day - April 20, 2020). The California Current Large Marine Ecosystem is outlined in red. The 200 m isobath is shown in white, and is a measure of the limit of the continental shelf. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

variable and 'boom-bust' recruitment (Baumgartner et al., 1992) with dramatic consequences for the fishery (Ueber and MacCall, 1992; Kuriyama et al., 2020). Sardines are important forage for the ecosystem (Kaplan et al., 2017), and show interannually variable migration and spatial distribution (Barange et al., 2009; McDaniel et al., 2016). Many studies link sardine dynamics to environment and climate conditions (Deyle et al., 2013; Jacobson & MacCall, 1995; Lindegren et al., 2013; McFarlane et al., 2002) although the relationships can be variable and the mechanisms unclear. There are two subpopulations of sardine defined for the CCS (Smith, 2005), and the northern subpopulation - distributed between northern Baja California and British Columbia - is the focus of U.S. management and our study. The southern subpopulation spawns in and typically inhabits the waters of Baja California, but can move into U.S. waters during warmer months. The habitat of the northern subpopulation has been linked to numerous environment and climate variables, especially sea surface temperature (SST) and surface chlorophyll (Zwolinski et al., 2011; Muhling et al., 2019). This subpopulation has a broad thermal niche, but increased presence when SST is 11–16.5 °C (Muhling et al., 2019).

Swordfish are a highly mobile and globally distributed oceanic species. They are caught in the eastern Pacific with large-scale and artisanal fishing gears including longlines, gillnets, harpoons and to a lesser extent recreational gear (IATTC, 2017). There is evidence of stock structure in the Pacific, with fish caught off California assumed to be part of the Western and Central Pacific stock (IATTC, 2017), but with some mixing with the Eastern Pacific Ocean stock in the Southern California Bight (Sepulveda et al., 2020). Swordfish distribution in the CCS is correlated with numerous environmental variables, including SST, surface mixing, and frontal features (Brodie et al., 2018; Scales et al., 2018), with greatest presence when SST is 16–20 °C. Swordfish can exhibit frequent vertical movement related to basking and foraging, and at night inhabit mostly surface waters (Sepulveda et al., 2018).

North Pacific albacore is considered to be one stock throughout the North Pacific, and spawning is restricted to tropical and subtropical waters in the Western and Central Pacific (Reglero et al., 2014). Albacore in the CCS are primarily immature juveniles that migrate during spring and early summer, and return to spawning grounds when mature (~5–6 years). Their distribution in the CCS is correlated with numerous dynamic variables, including SST, chlorophyll, and surface mixing (Zainuddin et al., 2008; Nieto et al., 2017; Muhling et al., 2019), and their seasonal- and age-based migration has been linked to the North Pacific transition zone (Polovina et al., 2001). Catches of albacore in the CCS are generally highest where SST is 15–20 °C (Muhling et al., 2019).

3.3. Key aspects of the socio-economic setting

CCS fisheries have important economic, social, and cultural value. The U.S. West Coast fishery system is characterized by its high levels of diversity, productivity, and variability, and CCS fisheries have long been an important contributor to coastal livelihoods, cultures, and economies (McEvoy, 1986; PFMC, 2022). In the modern era, many fishers across California, Oregon and Washington have historically participated in multiple fisheries, shifting effort within and between years in response to changes in species distribution and abundance, regulations, and market drivers (Kasperski and Holland, 2013; Fuller et al., 2017; Frawley et al., 2021, Fisher et al., 2021). Though access to marine resources was once largely unrestricted, limited entry licensing regimes, harvesting guidelines, spatiotemporal closures, and/or catch quotas were applied to many West Coast commercial fisheries in the 1970s, 80s, and 90s in order to combat overharvesting and to rebuild fish stocks. Following the passage of the Magnuson-Stevens Fishery Conservation and Management Act in 1976, the Pacific Fishery Management Council (PFMC) was created and charged with the management of the U.S. West Coast Exclusive Economic Zone (3-200 nautical miles offshore), development of relevant fishery management plans, and coordinating with individual states and the National Marine Fisheries Service (NMFS) to oversee their implementation.

Although a record number of regional stocks are now considered rebuilt or sustainably managed from a biological perspective (NOAA Fisheries, 2020), there are ongoing concerns about the social sustainability of the commercial fishing industry associated with the consolidation of access rights and fishing equipment (Russell et al., 2018), the "graying of the fleet" (Cramer et al., 2018), and ongoing environmental justice concerns related to the establishment of closed areas or gear constrictions (Mason et al., 2019). In recent years, climate change has emerged as a focal issue for fisheries stakeholders and management as shifting environmental conditions have resulted in changes in marine animal distribution and abundance and unfamiliar interactions between resource users, target species, seafood markets, and protected species (Richerson and Holland, 2017; Holland and Leonard, 2020; Santora et al., 2020). In recognition of this, the PFMC has developed the Climate and Communities Initiative (https://www.pcouncil.org/actions/climat e-and-communities-initiative), which is aimed at communicating the effects of climate change in the CCS among fishery managers and stakeholders, and to identify ways to incorporate such understanding into PFMC decision making.

Sardine in the CCS has an infamous history of high harvest and collapse in the early-mid 20th century. In U.S. waters, sardine are managed by the PFMC under the Coastal Pelagic Species (CPS) Fisheries Management Plan (Appendix A). Sardine are caught primarily with round-haul gear, and the same fleet will target other CPS species, including northern anchovy (*Engraulis mordax*), Pacific mackerel

(Scomber japonicus), and market squid (Doryteuthis opalescens). Sardine undergo regular stock assessment incorporating annual acoustic and trawl surveys, and are subject to a catch quota including a 'cutoff' abundance of 150,000 mt, which stops directed fishing when the subpopulation falls below this level. Due to low abundance, the sardine fishery closed in 2015, which coincided with a large marine heatwave and raised concerns about the vulnerability of this fishery to climate change.

Swordfish are important for commercial and recreational fishers in the CCS, and information on their redistribution is of management interest. Swordfish in the CCS are managed internationally by the Western and Central Pacific Fishery Commission (WCPFC) and the Inter-American Tropical Tuna Commission (IATTC), and in U.S. waters of the CCS by the PFMC which includes swordfish in its highly migratory species (HMS) Fisheries Management Plan (Appendix A). Swordfish in the U.S. waters of the CCS are predominantly caught commercially by the drift gill net (DGN) fishery. This is a small and declining limitedentry fishery and subject to extensive scrutiny due to bycatch issues (Mason et al., 2019; Savoca et al., 2020), and numerous regulations (such as time-area closures) are currently in place (Urbisci et al., 2016). One of the most scrutinized bycatch groups are sea turtles, and two spatially static closures (the large Pacific Leatherback Conservation Area, and the Loggerhead Conservation Area) exist strictly to reduce turtle bycatch.

The U.S. west coast commercial albacore fishery was historically important off California, but in the past 10-15 years has operated primarily out of Pacific Northwest (PNW) ports in Oregon and Washington. The fishery uses surface gear (hook-and-line and troll) and is not limitedentry, although there are some gear restrictions and vessels must hold a permit. Like swordfish, the North Pacific albacore fishery is managed internationally by the IATTC and WCPFC, and along the U.S. West Coast by the PFMC under the HMS Fisheries Management Plan (Appendix A). The stock is considered not overfished, and while the WCPFC has established a limit reference point, no formal harvest strategy or target reference points have been adopted (ISC Albacore Working Group, 2021). However, an international MSE process was recently conducted to assess performance of alternative management strategies and associated reference points to aid development of harvest strategy for North Pacific albacore by the two regional fisheries management organizations (ISC Albacore Working Group, 2021). There is potential for ongoing albacore spatial distribution shifts to impact U.S. landings, which may have broad impacts given the importance of albacore fishing in the harvest portfolios of West Coast fishers.

3.4. Stakeholder input

Fishery stakeholders were engaged from the onset of Future Seas to motivate specific research directions and strategically address management and fisher concerns. Engagement of stakeholders in research, especially research with management implications, is considered important for the uptake of model results and success of related management plans, and can enhance scientific understanding of complex systems (Mackinson et al., 2011; Aminpour et al., 2020,; Weiskopf et al., 2022). Direct stakeholder input occurred through two workshops and through semi-structured interviews (Appendix B). Semi-structured interviews have a set of questions but also allows new topics to be raised during the interview based on stakeholder answers (Bernard, 2017). Key concerns and themes which emerged from this engagement are summarized in Appendix B. Stakeholders included commercial fishers and industry representatives, fishery managers, fishery scientists, and nongovernmental organization representatives. The first workshop occurred within the first six months of the project (March 2018) and was focused on introducing the project and broad objectives to a variety of stakeholders across the three fisheries. The main goals were to get feedback on the project, discuss proposed methods such as management strategy evaluation, and identify key issues and priorities and related

management objectives that could be examined in the context of climate change. The second workshop in October 2018 had similar goals but focused specifically on CPS (especially sardine and squid). It was intended that stakeholder feedback and input be used to prioritize and guide our research. Key stakeholder concerns and themes included: change in target species distributions and abundance, fishery and ecosystem impacts of these changes, diverse issues related to changes in management strategies, and how well our modelling represents reality (Appendix B). While much of our research output does align with stakeholder concerns, we were not able to explore some desired issues in detail (e.g. sardine harvest strategies, squid abundance and distribution) due to limited data availability and resources.

4. Modeling framework and design

A modeling and analysis framework should be tailored to specific project goals, should be informed by the context of the project (Section 3), and should balance feasibility and realism. Also important is to capture key system and organism uncertainties in the design where possible. This task is not necessarily straightforward, so to aid similar projects we offer the example of how it was handled under Future Seas. We first describe the physical and biogeochemical modeling, and then for the ecological and fishery analyses we describe a series of intentional choices: (i) which models to use and why, (ii) how environmental information is incorporated, (iii) which management modeling framework is best suited to addressing stakeholder concerns, and (iv), what are the relevant metrics of change produced by the models. This article is not focused on technical details (which are reported in the cited studies or appendices) but more on the broad rationale. Details of the analyses and models, including data and code availability, are reported in Appendix C.

4.1. Physical and biogeochemical modeling

The structure of the ocean modeling was designed with consideration of several key needs and constraints: (i) resolving fine-scale physical and biogeochemical variability associated with coastal upwelling and mesoscale features, (ii) capturing uncertainty in potential future climates, (iii) resolving the evolution of ocean changes throughout the 21st century, and (iv) ensuring computational feasibility. Historical and future ocean conditions were simulated with a CCS configuration of the Regional Ocean Modeling System (ROMS). The ROMS domain covers 30-48°N and offshore to 134°W (Fig. 2), with a horizontal resolution of 0.1° (~10 km) and 42 terrain-following vertical levels (Veneziani et al., 2009). For the analyses described in this article, we use three types of model outputs based on this same ROMS domain. Each is described in detail in the associated references, so we provide only a short overview here (and in Table 1) and expand on the roles of different outputs in Section 7.3. First is a series of historical reanalyses (1980-2021; Neveu et al., 2016, oceanmodeling.ucsc.edu) – data-assimilative historical runs that provide our best estimate of the physical ocean state used for fitting ecological models. Second is a hindcast (1980-2010; Pozo Buil et al., 2021) - a realistic historical simulation without data assimilation but including biogeochemistry, which provides a baseline representation of the historical mean state against which future change can be quantified. Third is a set of three projections (1980–2100; Pozo Buil et al., 2021) which are forced by output from CMIP5 earth system models (ESMs) under a high emissions scenario (Representative Concentration Pathway (RCP) 8.5, Riahi et al., 2011). The three ESMs used to force projections -GFDL-ESM2M, HadGEM2-ES, and IPSL-CM5A-MR (hereafter abbreviated to GFDL, HAD, IPSL) - were chosen to span the range of physical and biogeochemical change in the CCS projected by the models in the CMIP5 ensemble. Model spread was prioritized over multiple scenarios as the former is the dominant source of physical and biogeochemical uncertainty in the CCS (Pozo Buil et al., 2021).

signals. In this approach, the changes in atmospheric and oceanic drivers projected by global climate models are applied to an observationally constrained estimate of contemporary ocean conditions, thus correcting regional biases common to global climate projections (e.g., Stock et al., 2011). Unlike many downscaling efforts (Drenkard et al., 2021), the Future Seas effort enlisted a 'time-varying delta' method that allows us to simulate the full 1980-2100 period (Pozo Buil et al., 2021). In contrast to the more common 'fixed delta' method that compares discrete future and historical periods, the time-varying approach allows us to provide the temporally evolving ocean conditions needed to study future ecosystem states that may reflect the integrated effects of conditions over several decades (Drenkard et al., 2021). Finally, the hindcast and projections are coupled to NEMUCSC - a CCS-specific version of the NEMURO biogeochemical model (Kishi et al., 2007; Fiechter et al., 2021). NEMUCSC includes three limiting macro-nutrients, two phytoplankton groups, three zooplankton groups, three detritus pools, and oxygen and carbon cycling. For computational efficiency, ROMS and NEMUCSC are run in successive steps (i.e. offline coupling), enabling a faster run time that is especially useful for biogeochemical parameter calibration and sensitivity studies.

4.2. Ecological and fishery modeling - Which models and why

Our models and analyses (Fig. 1) were selected to balance the project's aims and the priorities of the stakeholders and management against feasibility and flexibility. A key example of this balance was including population dynamics in models for sardine, but excluding this from swordfish models. Population dynamics or abundance was included in all sardine models due to the highly variable abundance of sardine - this is of great stakeholder interest, and excluding abundance would likely lead to less accurate models of their distribution. For swordfish we felt confident focusing on high resolution spatial models (which are more relevant to current management priorities) while excluding population dynamics because the status of swordfish stocks in the North Pacific and CCS is relatively stable and not overfished (ISC Billfish Working Group, 2018) and because processes such as recruitment occur outside the CCS. Modelling population dynamics at a fine spatial scale is a great challenge, so it is common in projections to focus more on spatial distributions or more on abundance trends.

Given the focus of stakeholders on questions of future environmental change and spatial distribution shifts, we developed and projected correlative species distribution models (SDMs) for sardine (Muhling et al., 2020), swordfish (Appendix D), and albacore (Appendix E). These SDMs were then linked to correlative landings models to project the potential impact of distribution shifts on fishery landings for both sardine (Smith et al., 2021a) and albacore (Appendix E). The albacore model also leveraged a population dynamics model used in the recent North Pacific albacore MSE (ISC Albacore Working Group, 2021), to ensure that the impact of changes in biomass (driven by management strategies or productivity) could be linked to changes in the availability and community landings derived from the SDM. The sardine and albacore landings models assess potential consequences of climate driven shifts in habitat for community-level landings to enable managers to better understand climate risk and develop strategies to increase resilience of affected communities. We also developed an analysis that evaluates SDM performance to measure the timescales of ecological predictability (Brodie et al., 2021), which allows us to better identify the scales of environmental covariates that should be prioritized in SDM development.

Additional priorities for stakeholders and managers in the swordfish fishery are bycatch and spatial management, so we linked correlative SDMs with an economic model to evaluate how distribution shifts can affect the opportunity costs of spatial closures (Smith et al., 2020), and used SDMs to provide the ecological dynamics in a complex MSE model to evaluate the effectiveness of static and dynamic spatial closures for the management of bycatch (predominantly turtles) in the DGN fishery (Smith et al., 2021b). The MSE also included an agent-based model to simulate fishing effort, which was deemed essential given the need for flexible responses of simulated fishers to dynamic closures. Projecting the MSE was deemed non-essential given the additional uncertainty of projecting the agent-based model and simulating the associated fishing effort, and because the management implications given future species redistribution could be inferred from historical patterns and interannual variation.

To explore the potential future abundance and distribution of sardine, we projected an individual-based model (IBM) of sardine coupled to ROMS and MEMUCSC (Rose et al., 2015; Fiechter et al., 2021), as well an ensembled age-structured population dynamics model (Koenigstein et al., 2022). Alongside the SDM-based landings projection, these two models provide diverse and comparable approaches to exploring the future of sardine. All three models were projected to 2100 based on output from the three downscaled projections, allowing us to evaluate uncertainty associated with both climate and fishery models. The IBM is a complex framework that can incorporate individual-level processes and is entirely mechanistic and operates at a fine temporal and spatial scale, whereas the environmental-informed population dynamics model can also incorporate mechanism but is more coarsely resolved, but does allow for more rigorous fitting to fishery-dependent time series.

And finally, due to broad concerns about availability of fish to fisheries, and in the case of albacore stakeholder concerns relating to the West Coast fishery's 'open access' status, we developed models and analyses to explore community engagement and fishery structure and participation. The community and processor engagement analysis uses dimension reduction to measure the relative involvement of communities in a fishery (e.g. Himes-Cornell and Kasperski, 2016; Appendix F), which is useful for managers as it can identify the most relevant areas for risk assessment and management. This analysis prevised a metric similar to the social indicator used in NOAA's ecosystem status reporting (Colburn et al., 2017; Harvey et al., 2020). The analysis of albacore structure and participation allows more insight into how a fishery is structured, and provides an assessment of longitudinal changes impacting the fishery system and the fishery's trajectory. This uses a broad 'methodological triangulation' analysis, detailed in Frawley et al. (2021).

4.3. Ecological and fishery modeling – How environmental information was incorporated

For the SDMs, environmental information was included as habitat correlations, predicting the spatial distribution as probability of presence of sardine (Muhling et al., 2020; Smith et al., 2021a), the probability of presence of swordfish (Appendix D) or the catch rate of swordfish and key bycatch species in the DGN (Smith et al., 2021a; Smith et al., 2021b), and the catch rate of albacore (Appendix E). Of note was the addition of a 'sardine abundance' covariate in the sardine SDM, which can allow for abundance-dependent habitat associations (Muhling et al., 2020). For the SDM-based projection of albacore landings, we also incorporated a transition zone index directly into the landings model (Appendix E), as albacore are known to use the North Pacific transition zone (the region where the North Pacific subtropical and subpolar gyres meet) as a migration pathway to the West Coast. This landings analysis can be enhanced for consideration of environmental effects on productivity either implicitly by using scenarios of future productivity trends (i.e. in recruitment, mortality, and growth; e.g. Punt et al., 2016) or explicitly by directly integrating an environmental index into the population model (e.g. Haltuch et al., 2019). The SDM-based MSE for swordfish additionally includes the environment in the assessment model, where environmental information is used to build spatial closures (Smith et al., 2021b).

In the IBM and population dynamics models of sardine, the environment is included mechanistically, which allows the exploration of

causal relationships and production of outputs other than spatial distributions, although functional forms and parameter values can be more challenging to estimate. In the IBM, environmental drivers (primarily temperature and zooplankton prey) influence several properties of sardine individuals, such as growth, survival, reproduction, and movement behavior (Rose et al., 2015; Fiechter et al., 2015). Zooplankton concentrations are determined from the NEMUCSC component of the coupled model. In the population dynamics model, SST determines spawning location and early life stage survival, nanophytoplankton and microzooplankton biomass (also from NEMUCSC) determine late larval and juvenile survival, and adult sardine consumption and egg production are influenced by diatom, mesozooplankton, and krill biomass (Koenigstein et al., 2022). Environmental information from the ROMS reanalysis was used to fit all the SDMs, while the ROMS-NEMUCSC hindcast was used to fit and tune the IBM and population dynamics models. The downscaled ROMS-ESM projections were used to drive sardine projections and compare the state of the sardine population under historical and future periods (see Section 7.3 for a discussion of integrating historical and future products, and a recommended workflow).

The community engagement analyses for albacore links port-level landings with vessel and processor spatial information. These analyses do not incorporate dynamic environmental variables, and nor does the fishery structure analysis of the albacore fishery. These types of analyses are not designed for incorporation of environmental variables; however, these analyses provide a means to infer potential broader impacts of spatial distribution shifts on the fisheries.

4.4. Ecological and fishery modeling - Management modeling framework

Of keen interest to stakeholders are projections of landings, and both our sardine and albacore landings projections needed to make assumptions about highly uncertain future fishery characteristics, such as the location of ports, fishing effort, management regulations, and the abundance of other target species. For sardine, the SDM-based simulation, the IBM, and the population dynamics model approach this differently. In the SDM-based simulation, future uncertainty is addressed by assuming all future conditions - except for sardine habitat and their spatial distribution - are the same as those in an historical reference period (Smith et al., 2021a). The structure of the linked landings model is such that this analysis also evaluates whether the current seasonal quota allocation scheme, or the landings of other CPS, might constrain future landings. In the IBM, an agent-based fleet model simulates fishing originating from five west coast ports, with daily fishing effort determined by accessibility to nearby sardine individuals and expected revenue from catch. And in the age-structured population dynamics model, accessibility of the stock to the California and PNW fleets is determined by its dynamic spawning location, while total landings are governed by a statistical relationship of yearly fishing quota to stock abundance in the previous year. The albacore landings projection is similar to the SDM-based analysis of sardine, but with some key differences driven by species and fishery differences. For example, albacore availability to West Coast vessels are impacted by environmental conditions outside our ROMS domain, and albacore availability is linked to vessel size. Thus, the model was stratified by vessel size, and the center of gravity (COG) was used as a more robust measure of albacore distribution and availability (Appendix E). Like sardine, future fishing effort was based on a sample of historical effort, in order to avoid unrealistic extrapolation.

Analysis of management strategies for swordfish involved a risk assessment of an existing spatial closure (the Loggerhead Conservation Area; Smith et al., 2020), a projected risk assessment of existing spatial closures, and an MSE comparing static and dynamic spatial closures (Smith et al., 2021b). These studies explored strategic issues related to the sensitivity of static management in a dynamic and changing ocean. The projected risk assessment incorporated climate projections from the three ESMs to 2050, but the Loggerhead Conservation Area risk assessment and the MSE were historical and used the ROMS reanalysis for 1991–2009 and 1991–2000, respectively. The projected risk assessment of existing spatial closures allowed us to evaluate the potential impact of existing spatial closures on DGN fishers given a potential climate-driven redistribution of swordfish. However, to allow a tractable analysis, we assumed a constant swordfish price and fishing costs, and we ignored the Loggerhead Conservation Area, due to the challenge of predicting future El Niño years (Welch et al., 2019).

Our analysis of long-term changes to the structure and function of the albacore troll and pole-and-line fishery (Frawley et al., 2021) used participation networks (Fuller et al., 2017). This method enabled us to describe social-ecological linkages and feedbacks that inform how U.S. West Coast fishers participate and shift effort among fisheries. By exploring the consequences of overlapping changes in albacore ecology, seafood markets, and fishery regulations for diverse user groups, we identified patterns and processes likely to mediate the distribution of economic and non-economic fishery benefits in response to future changes in climate and/or governance.

4.5. Ecological and fishery modeling - Key metrics

For the SDMs, the key metrics are typical of these models: 'probability of presence' for models created from presence-absence data, and 'catch rate' for those created from fishery-dependent catch data. These are metrics of 'habitat suitability', which is generally the expected probability or abundance of a species in a given habitat. These catch rate models account for fishing effort, which becomes an input in SDM predictions. COG is also a key metric as it summarizes a general change in a finely resolved SDM. The SDM-based landings simulations of sardine and albacore ultimately quantify (respectively) percentage change in landings due to sardine spatial redistribution (Smith et al., 2021a), and the change in landings per vessel (Appendix E). Additional metrics evaluated for the SDMs were model predictive performance under anomalous environmental conditions (Muhling et al., 2020), and model predictive performance that can be attributed to each of the decomposed elements of SDM covariates (i.e. climatology, low frequency signal, high frequency signal; Brodie et al., 2021).

The metrics evaluated in the IBM and population dynamics models reflect their additional complexity. For the IBM, they include the historical and future sardine population biomass, spatial distribution of adult sardines and eggs, and regional annual catch. In the population dynamics model, metrics include stock abundance and age structure, landings by two fleets, latitudinal distribution shift of the stock, and the contribution of different biological processes to increases or decreases during the projection. Due to the focus on spatial management and bycatch, key metrics for the swordfish MSE and risk assessment are overlap of swordfish with closures (which as an economic metric becomes 'lost opportunity'; Smith et al., 2020), rates of catch and bycatch, and the bycatch:catch ratio (Smith et al., 2021b; also called the reduction ratio, Dunn et al., 2016).

The community and processor engagement analysis for albacore uses factor analysis to reduce dimensions of input variables to engagement indices (Himes-Cornell and Kasperski, 2016), which in our study are a commercial harvesting index and a commercial processor index. Indices are derived for each community with documented historical participation in each fishery, based on the number of participating vessels owned, volume of the species landed by participating individuals, and ex-vessel value of that landed volume (for the harvesting index); or derived from the number of participating processors, volume landed at these processors, and the ex-vessel value of that landed volume (for the processor index). Note that the harvesting index is based on where vessel owners live (not necessarily where landings are made), and the processor index is based on the location of landings irrespective of where the vessels are from. Further detail of this method can be found in Appendix F.

For the longitudinal analysis of structure and participation for the

albacore fishery (Frawley et al., 2021), the key metrics evaluated were changes in total catch and effort, the proportion of active fishing vessels targeting albacore, the relative diversification of different fleet segments targeting albacore, and the node strength and centrality of albacore in the participation networks of different fleet segments.

5. Results

Key results and uncertainties are summarized in Table 2 and detailed below. Uncertainties in Table 2 include those results which vary among ESMs, are unlikely to be precise due to (for example) being difficult to parameterize, or are key unmodelled aspects.

Table 2

Summary of main results for each reported model or analysis. Key uncertainties are the results that may not be well resolved or that differed among ESMs.

Model/Analysis	Key results	Key uncertainties
ROMS-NEMUCSC projection Climate envelope analysis	Increased surface warming, stratification, and subsurface nitrate enrichment and deoxygenation Climate novelty in the CCS increases from 5 to 10 % in 2040 to almost 100 % by 2090	Rate of warming; changes in conditions (especially nutrients and oxygen) in coastal areas Rate of warming greatly alters timing and magnitude of novelty
Sardine SDM	Sardine habitat shifts northward	Rate of habitat shift; suitability of near-shore habitat; potential for non- stationarity in responses (Muhlinget al., 2020)
Sardine SDM-based landings projection	Sardine landings shift northward (higher at northern ports, lower at southern ports)	Amount of change in total landings; whether landings of southern sardine subpopulation will increase
Sardine IBM projection	Sardine habitat and landings shifts northward; Long-term abundance relatively stable with increasing from 2070	Magnitude of high- frequency abundance change not well modelled; end-of-century population state
Sardine population- dynamics projection Swordfish SDM	Sardine habitat and landings shifts northward; abundance recovers from current lows Swordfish habitat suitability generally increases, and shifts northward	Magnitude of abundance change due to model configuration Predictability of fine scale habitat suitability (Brodieet al., 2021)
Swordfish spatial closure 'lost opportunity' projection	Swordfish distribution shifts unlikely to change impact of current spatial closures	Future change in fishing gear and subsequent bycatch patterns
Swordfish spatial closure MSE	Dynamic closures often more effective than static, but create practical challenges for fishers and managers	The occurrence of rare species within their suitable habitat
Albacore SDM	Albacore habitat shifts north and potentially inshore	The continuation of a migration corridor for albacore to reach the CCS
Albacore landings projection	Landings decrease in Southern California but increase or are stable elsewhere; fishing effort increasingly straddles the U. S./Canada border	Rate of warming, which impacts magnitude of landings change (but not spatial patterns)
Albacore engagement analysis	Only some southern communities engaged in harvesting likely to be affected by projected albacore range shifts	How communities and ports respond to shifts in albacore
Albacore participation network	Albacore supports diverse harvest portfolios; the success of small-boat operations appears contingent on their ability to opportunistically target albacore in coastal waters	How fishers respond to potential management changes

5.1. Physical and biogeochemical change

Results from the ROMS-ESM downscaled projections indicate some common changes of the CCS properties across ESMs. These include an intensification of upwelling-favorable wind stress in the northern CCS, an overall surface warming and increased stratification, and subsurface nitrate enrichment and deoxygenation (Fig. 3). However, there are clear differences in future properties among ESMs, especially in coastal regions for variables such as NO₃, O₂, and chlorophyll (Fig. 3). Here we highlight some of the key results from the ROMS-ESM projections, with a more comprehensive analysis provided in Pozo Buil et al. (2021).

5.1.1. Future changes in physical properties

All three downscaled projections show an overall intensification of the meridional wind stress (i.e., proxy for upwelling favorable winds) in the northern CCS, with ROMS-IPSL showing the weakest intensification and ROMS-GFDL the strongest. In contrast, projected changes in the

southern CCS differ between models (Fig. 3). Surface warming is consistent across all models, though its magnitude is region- and modeldependent. By the end of the century, ROMS-GFDL projects the weakest increase in SST ($\sim 2^{\circ}$ C), with weaker warming in the northern CCS, while ROMS-IPSL and ROMS-HAD project stronger increases in SST (~4°C and ~ 4.5 °C, respectively). These temperature trends in the downscaled projections closely follow those of their parent ESMs (Fig. 3). Warming in the CCS is surface intensified, leading to enhanced stratification (Pozo Buil et al., 2021; Cordero-Quirós et al., in review). This has further implications for the physical ocean environment: first, a projected reduction in mixed layer depth is evident in all three models (Bograd et al., 2023); second, a more stratified ocean drives a more energetic mesoscale field, and all three projections show enhanced eddy kinetic energy (EKE) with values at the end of the 21st century that are up to five times those from the 1980-2010 historical period (Cordero-Quirós et al., in review).



Fig. 3. Trends in five example physical and biogeochemical variables from the three downscaled projections; the rows from top to bottom: meridional wind stress, sea surface temperature (SST), nitrate concentration (NO_3) at 150 m, dissolved oxygen (O_2) concentration at 150 m, and 0–50 m vertically integrated chlorophyll concentration. *Left panels*: maps of mean absolute values from the historical control run (1980–2010; first column), and mean anomalies for a future period (2070–2100, relative to historical). Negative values of wind stress represent an increase in equatorward wind stress. Black contour marks indicate 100 km from shore. *Right panels*: time series of yearly averaged anomalies (relative to historical), averaged from the coast to 100 km offshore for the same five variables, and averaged for the northern (>39°N, left column) and southern CCS (<39°N, right column). Results from downscaled and global model resolutions are compared (e.g. ROMS-GFDL vs GFDL).

5.1.2. Future changes in biogeochemical properties

In the coastal regions, ROMS-GFDL and ROMS-IPSL project similar biogeochemical responses of \sim 30–40 % average increase (decrease) in subsurface nitrate (dissolved oxygen) by end of century. ROMS-HAD projects an opposite response featuring a decrease (~20 %) of subsurface nitrate along the coast and an increase (~30 %) of subsurface dissolved oxygen in the southern and central coastal waters. The evolution of these anomalies in ROMS-IPSL diverge in magnitude from those in the IPSL ESM, especially in the southern coastal region (Fig. 3). The most pronounced projected changes in chlorophyll occur in a coastal band between 35 and 45°N, the most productive portion of the CCS. ROMS-GFDL projects a weak increase of upper ocean (top 50 m) chlorophyll in the northern coastal region ($\sim 3 \text{ mg m}^{-2}$), whereas ROMS-IPSL and ROMS-HAD project moderate (-0.5 mg m^{-2}) and strong (-10 mg m^{-2}) declines, respectively, by the end of the century. Chlorophyll concentrations in the downscaled projections qualitatively track those of their parent ESMs, but the magnitude of change can be amplified (ROMS-IPSL) or reduced (ROMS-HAD) in the downscaled models. (Fig. 3).

5.1.3. Emergence of novel climates

A climate envelope analysis measures change in a multivariate space define by numerous climate variables, and novel (future) climates occur when climate conditions exceed the bounds statistically defined for a comparison (historical) climate (Williams et al., 2007; Mahony et al., 2017; Smith et al., 2022). Rather than focusing on change in single variables (e.g. SST), a climate envelope analysis examines multiple variables simultaneously, recognizing the importance of novel combinations of variables. Climate novelty can: 1) reveal novel environmental conditions otherwise difficult to identify, and 2) provide a general indication of environmental stress for a broad range of species. When considering the envelope defined by key drivers of fisheries habitat (i.e., SST, dissolved oxygen, mixed layer depth, and eddy kinetic energy), we find that (compared to the 1980-2009 period, and under RCP8.5) consistent novelty in the CCS climate envelope appears first in \sim 2040 in small patches off Southern California and the PNW, but by 2060 about 50 % of the CCS could experience a novel climate and almost all the CCS could experience a novel climate by the end of the century (Smith et al., 2022; Fig. 4). This is for mean monthly conditions, which represents a more persistent level of novelty than short-term heatwaves. The main driver of climate novelty is ocean warming, but weaker contributions also come from lower dissolved oxygen (especially inshore) and shallower mixed layer depth (especially offshore). There is great potential in exploring species- and model-specific projections of novelty using

climate envelope analysis, with the primary focus being on substituting or adding other variables (e.g. bottom temperature, pH, nutrients) to reflect different aspects of an ecosystem (Smith et al., 2022).

5.2. Ecological and fishery change

5.2.1. Sardine

The preferred habitat of the northern subpopulation of sardine is projected by an SDM to shift northwards, with offshore areas becoming generally more unsuitable (Smith et al., 2021a). The IBM and population dynamics model show a similar trend, with the IBM estimating a poleward shift of 500–800 km during the 21st century, depending on the rate of warming (Fiechter et al., 2021). However, there is evidence that sardine have non-stationarity in their habitat associations, which was particularly evident during recent marine heatwaves (Muhling et al., 2020). This non-stationarity might be due to a mismatch between environmental conditions and spatial–temporal cues (such as migration or spawning), and suggests that projections based purely on correlative SDMs should be interpreted cautiously.

Projections from the IBM showed substantial low-frequency variability in sardine population biomass, with a decrease in 2020-2040 and strong increase beginning in the 2070s (Fiechter et al., 2021). In the agestructured population dynamics projections, sardine abundance moderately increases over time, with significant fluctuations caused by underlying environmental variability. Under ROMS-GFDL the population dynamics model shows an interim period of low sardine biomass in the middle of the projection due to low planktonic food availability in combination with a temporary warming hiatus (which may indicate a period of low ecological productivity), while the high temperatures reached towards the end of the century under ROMS-HAD and ROMS-IPSL drives a population increase to historical peak levels, but also high ecological uncertainty (i.e. divergence among model ensemble configurations). Both the IBM and population dynamics models show that spawning areas shift far northward in the CCS towards the end of the century, while feeding areas shift only moderately northward. This discrepancy could lead to a change in sardine migration patterns, which would reduce the impacts of surface ocean warming on the sardine stock and on accessibility to fisheries.

Across the three ESMs, the SDM-based simulation projects a decline in sardine landings of 20–50 % at Southern California ports by 2080, and an increase of 0–50 % at PNW ports (Smith et al., 2021a; Fig. 5). The seasonality of the fishery was also projected to change, with the duration of the fishing season generally lengthening, but constrained by landings



Fig. 4. Maps of locations in the CCS with novel or analog projected climates. Maps are shown for August at the end of five decades, under ROMS-IPSL. Colors represent an analog climate (green), or three scales of climate novelty (yellow, orange, red). The maps show each grid cell's majority classification over a 5-year period ending in the specified year (e.g. 2016–2020). Novelty was calculated using the hypervolume method, based on a climate envelope of SST, dissolved oxy-gen, mixed layer depth, and eddy kinetic energy (Smithet al., 2022). The black line is the exclusive economic zone (EEZ). (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)



Fig. 5. Projected values for sardine latitudinal center of gravity (COG, a–c) and sardine annual landings (c–e) for the three models (IBM, age-structured population dynamics [Pop-dyn], SDM simulation) and three ESM projections (black is ROMS-GFDL, blue is ROMS-IPSL, red is ROMS-HAD). Landings are also shown as percentage of total landings in the California and Pacific Northwest (PNW) areas (f–k). In the IBM and SDM simulation landings were aggregated by summing port-level landings. For the population dynamics model, the mean of nine ensemble model configurations is shown, and the modeled period begins during a state of fishery closure (below 150,000 tons; d). Because the three models differ in how (or if) they model effort, sardine biomass, subpopulation structure, and fishery closures, COG (a–c) and the relative change in California and PNW landings (f–k) are the most robust results to compare across models. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

of other species and the seasonality of the quota allocation. The IBM projects change in catches due to changes in both sardine distribution and abundance. Catch is projected to increase by 50–70 % in the northern CCS and decrease by 30–70 % in the southern and central CCS. The late-century increase in sardine abundance in the northern CCS exhibited a large spread across climate projections, which suggest that end-of-century results bear a substantial level of uncertainty. The population dynamics modeling differs by beginning the projection period in a state of fishery closure, representing the current state of the actual fishery. After the modeled fishery reopens, the ensemble indicates a moderate increase in total sardine landings in the long term, while interannual and interdecadal variability remains high. Landings in the PNW increase in significance in comparison to California in the middle and end of the century (Fig. 5).

The three modeling approaches agree on the general future trends in sardine spatial distribution (a $\sim 4^{\circ}$ latitude shift northward by 2100) and its impact on landings, with the California fleet most vulnerable to a shifting distribution (Fig. 5). However, the uncertainty within models and among ESMs was high, and similar to the uncertainty in relative change among the three modeling approaches. To develop effective adaptation strategies for sardine, it will thus be important to assess

performance of current and alternative harvest controls, quota allocation schemes, and monitoring systems given the projected shift in distribution and climate and ecological uncertainty. While an MSE was not attempted for sardine in the first phase of the Future Seas project (but see Section 6, Future directions), the SDM-based simulation did investigate how the current management strategy might constrain future landings given a future range shift (Smith et al., 2021a). This simulation shows that a seasonal quota allocation can limit landings when the availability of the species shifts temporally; i.e. to maximize adaptation to the projected range shift, the future seasonal allocation should be sensitive to a changing fishing season phenology. The simulation also indicated the importance of considering the futures of other CPS and fisheries impacts from a CPS guild perspective, as the timing of other CPS landings constrained sardine landings even when sardine presence was high. Finally, this work stresses the importance of monitoring and modeling the distribution of the southern sardine subpopulation, which may increasingly migrate into U.S. waters and potentially ease projected negative impacts on the California CPS fleet.

5.2.2. Swordfish

Swordfish habitat in the CCS is broad but sensitive to SST in

particular, and projections show a poleward and offshore redistribution of habitat, with an average 52 % increase in total habitat suitability (Fig. 6). The magnitude of the COG shift from the historical distribution range (1985–2015) to the future distribution range (2070–2100) is an average of 198 km northwards and slightly offshore, with only small variation among ESMs. Habitat suitability decreases only in nearshore areas (Fig. 6).

Models of catch rate show the strong influence of SST and MLD, which can mean a strong seasonality of (and potential for long-term change in) the inshore-offshore distribution of swordfish, which may also change coincident with Northeast Pacific climate regimes (Smith et al., 2020). The analysis of the timescales of predictability of swordfish shows that a monthly climatology is sufficient to explain much of the swordfish distribution at these scales, and the addition of low frequency (interannual) environmental variability provides most of the capacity to predict anomalous distribution and catch, whereas high frequency variability adds only minor improvement (Brodie et al., 2021). This result indicates that catches of swordfish in the DGN fishery are more predictable at broader spatial and temporal scales, with uncertain association with finer scale ephemeral ocean features. This gives confidence in distribution projections for highly migratory species, as broadscale distribution patterns are primarily driven by low frequency climate variability - a timescale that is better captured in ESMs compared to fine scale variability.

Given the likely redistribution of many species in the CCS, including swordfish, the spatial management of bycatch in the swordfish DGN fishery will benefit from increased dynamism (Pons et al., 2022), i.e. closures that can change in time or space. Even without a climate trend, interannual variation in ocean environment and species' distribution can have a large proportional impact on a static closure's impact (Smith et al., 2020). When this historical analysis is extended to project 'lost opportunity' out to 2050, we see evidence that a redistribution of swordfish leads to changes in fishing opportunity (the proportion of the profitable fishing area outside closures), but that there is likely no change, or even an increase, in future fishing opportunity (Fig. 7). In other words, the economic impact of the existing static closures on the DGN fishery is unlikely to increase due to any climate-driven changes in the spatial distribution of swordfish; where 'economic impact' here relates to the severity of changes to effort or fisher behavior that will occur due to a closure's enactment (Smith et al., 2020). However, opportunity is just one metric for evaluating spatial closures. Our MSE highlights that the key to the continued success of DGN spatial closures for bycatch reduction will be following the future distribution of sea turtles, and the evaluation of more dynamic closures (alongside other bycatch

mitigation tools) that match the dynamism of the species' distribution (Smith et al., 2021b; Kaplan et al., 2021). New gear types that are being tested to catch swordfish with lower bycatch rates may provide an opportunity as swordfish change in distribution and abundance in the future. Dynamic closures can achieve similar or better reduction of bycatch compared to static closures (Pons et al., 2022), while allowing higher target species catch and fisher profit (Appendix G).

5.2.3. Albacore

Projected COGs from the albacore SDM move generally inshore and northward in the future (Fig. 8a). In the CCS, predicted CPUE decrease in the southern and offshore study region, but increase closer to shore north of Point Conception (Fig. 8b). These shifts are due to SSTs becoming warmer in the offshore CCS than those historically linked to albacore fishing grounds, but becoming more favorable in the nearshore upwelling zone. Historically, albacore catches have been low in the high-chlorophyll region close to shore, but a combination of warming and reductions in surface chlorophyll nearshore – projected by the ROMS-IPSL and ROMS-HAD models – increase the favorability of this area for albacore in the future. However, the ability of albacore to make use of this increasingly nearshore suitable habitat may also depend on continued availability of a suitable migration corridor to reach the CCS.

The distribution of albacore fishery landings and effort has been highly variable across space and time (Frawley et al., 2021). The community engagement analysis shows that numerous communities along the West Coast are currently engaged in the albacore fishery, with location-based processing engagement centered in the PNW, and harvesting engagement more uniform along the coast (Fig. 9a). Most communities have very low engagement in the fishery, with a handful dominating both types of engagement, but especially processing. A community is considered highly engaged when engagement score is considerably higher than the mean score for all involved communities (see Appendix F). Vessel size influences engagement, with the most communities engaged for small vessel harvesting or processing, and the least for large vessels. Processing engagement otherwise shows little difference among vessel sizes, while for harvesting engagement communities in Central California show more engagement with small vessels, and Southern California with large vessels. Albacore has become an increasingly important component of the harvest portfolios of smallboat fishermen (Frawley et al., 2021), many of whom lost access to other traditionally important species assemblages due to environmental change and regulatory reform. While large high-volume vessels based in Southern California have been able to accommodate significant spatial shifts in the fishery, the success of range-restricted and capacity-limited



Fig. 6. (a) Maps of change in the habitat suitability of swordfish and their center of gravity (COG), for the three ROMS-ESMs. Color represents change in the probability of presence between historical (1985–2015) and future (2070–2100) periods (red an increase, blue a decrease). Change in bivariate COG is represented by historical (green) and future (purple) bars. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)



Fig. 7. Results of the 2010–2050 projection of fishing opportunity in the DGN. (a) An example date (30-Sep-2030) of estimated utility (the difference between swordfish revenue and fishing costs) given swordfish catch-*per*-unit effort on this day predicted by an SDM. The red areas are those with profitability (positive utility) for a vessel leaving San Diego (red dot) on a 3-day fishing trip and landing at the nearest port (see <u>Smithet al.</u>, 2020). There are seven closures active on this date (five possible in this extent; hashed areas and purple lines) including the large Pacific Leatherback Conservation Area (PLCA). (b–c) The fishing opportunity (proportion of profitable area outside closures) is shown for every year from 2010 to 2050 for the extent of the fishing season (Jul-Jan), with the last 10 years (2041–2050) shown as black lines (SDM1) or red lines (SDM2). Two plausible SDMs (SDM1 and SDM2; based on those in <u>Smith et al.</u> (2020)), were used to account for structural uncertainty. (d) The changes in opportunity (b–c) can be summarized as the mean for each fishing season of daily total profitability (the sum of the positive utility in open areas). We see that SDM1 (black lines) indicates fishing opportunity will be stable to 2050, while SDM2 (red lines) indicates fishing opportunity will increase. (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

small-boat operations appears contingent on their ability to opportunistically target albacore in coastal waters and to maximize catch value as part of a diverse harvest portfolio.

The albacore landings projection demonstrates that different fisher groups have different strategies for adapting to changes in albacore distributions, and that communities in southern California are likely to be negatively impacted by future change, while communities in the PNW may benefit. Larger vessels are projected to be more responsive to changes in albacore distribution, employing a 'follow the fish' adaptation strategy, and switching landing ports more quickly if it becomes economically beneficial. Small vessels, which have more limited mobility, are more likely to keep landing in the same community. As the COG of albacore distribution moves northward (Fig. 8a), projected landings from large and medium vessels in 2041-2070 and 2070-2100 show a 25-100 % increase in landings in PNW communities and a 5-60 % decrease in San Diego and Long Beach relative to average landings in 1995-2018 (Fig. 9b). Small and medium vessels rely on a more diverse portfolio (Frawley et al., 2021), and, unlike for large vessels, projected changes in landings depend on biomass trends of other species and estimated albacore availability within 60 miles (small vessels) or 120 miles (medium vessels) from port. Variability in considerable among ESMs (Fig. 9b) due to the variability in nearshore habitat suitability, with ROMS-HAD showing higher suitability than ROMS-IPSL or ROMS-GFDL (Fig. 8b). Our results are consistent with those of Phillips et al. (2014), who found a positive but spatially variable effect of SST on historical albacore catch rates and hypothesized that climate change would negatively affect southern albacore-reliant communities, while northern communities would be positively impacted.

Our social-ecological investigation of the albacore fishery suggests that the fishery is composed of diverse participants who may be differentially impacted by projected changes to the CCS. While some ports, processors, and vessels classes have demonstrated the capacity to negotiate large-scale shifts in distribution and abundance, the negative impacts upon others are likely to be significant. This may be particularly true for processing communities in Southern California that are heavily dependent on North Pacific albacore. Given albacore's role in helping to support diverse harvest portfolios and promoting operational flexibility, changes impacting the albacore fishery may have cascading impacts upon the effort allocated to other linked fisheries (i.e. salmon and Dungeness crab). More broadly, our analysis of divergent impacts and responses across albacore fishery user groups suggests the limitations of a management strategy that applies equally to all fishery participants. The design of equitable and effective interventions may necessitate explicit accounting of the distinct response capacities and vulnerabilities of different user groups. This consideration is important in light of recent interest by regional fisheries management organizations to establish a harvest control rule for North Pacific albacore and the recent MSE (ISC Albacore Working Group, 2021). Our landings model adds to the performance metrics used in the MSE by providing a framework with which to test impacts of different HCRs on metrics reflective of the heterogeneous nature of fishery participants. The model also highlights factors affecting the catchability of the U.S. albacore fleet, which can inform development of climate-driven implementation error scenarios for this fleet. Finally, projected albacore fishing effort more often straddles the U.S./Canada border by 2100 (Fig. 8a), highlighting the need for continued coordination of fishing access across national jurisdictions via



Fig. 8. (a) Projected latitudinal center of gravity (COG) of fishing effort for North Pacific albacore in the Eastern Pacific Ocean for the three ROMS-ESMs. Points represent multiple iterations of each year based on resampling of historical fishing locations. Points are also colored by year. (b) Maps of projected mean albacore habitat suitability in 2080–2100 shown as change in CPUE [log10(x + 1) fish/vessel/day] relative to 1990–2020 (i.e. an anomaly) from the albacore SDM for the three ROMS-ESM projections (red is better habitat, blue is poorer habitat). (For interpretation of the references to color in this figure legend, the reader is referred to the web version of this article.)

the existing United States-Canada albacore treaty.

6. Future directions

The first phase of the Future Seas project has focused on highlighting patterns of future change and potential impacts to fisheries socialecological systems, and informing strategic decision making for fisheries management. This information can be used in scenario planning, by providing a range of potential environmental change as well as plausible future fish distributions and fishing grounds. Our results can also help prioritize tactical research; for example, we identified the sensitivity of the sardine fishery quota allocation strategy to sardine redistribution, which indicates the value of an MSE exploring specific alternatives. Likewise, our swordfish MSE provides further evidence that dynamic spatial closures are valuable for bycatch mitigation, but that managing very rare species comes with high uncertainty. This result could spur development of an MSE for specific closure types in the DGN (rather than the generic DGN-like fishery described herein), including hybrid strategies (e.g. incorporating hard caps) that can help manage the increased level of bycatch uncertainty. Our long-term projections, and evaluation of current and historical fishery structure and engagement, can also feed into NOAA's Integrated Ecosystem Assessment program (Harvey et al., 2017) by contributing to risk assessment of fisheries and species, and quantitative scenario analysis in the CCS (Holsman et al.,

2017; Samhouri et al., 2019). By focusing in detail on specific fisheries, our community engagement analyses complement the broader suite of metrics available in NOAA's Community Social Vulnerability Indicators Toolbox (Jepson and Colburn, 2013), which identify fishing communities susceptible to the adverse impacts from a range of stressors (Kasperski et al., 2021).

The second phase of Future Seas is underway, and aims to develop and improve our modeling framework. To date, Future Seas has used predominantly-one-way coupling between ecological and social components, i.e. the environment and species were not affected by the social system (Kasperski et al., 2021). In general, the social components of our models are less well developed than the environment and ecological components. Thus, our goals for the future include: enhanced two-way coupling, including feedbacks between fishing and species' abundances and distributions; more tactical environment-informed MSE models testing harvest control rules; improved integration of the environment into biological models (for example, via recruitment in population dynamics models); and more comprehensive social system components. Enhanced dynamic feedback between social and ecological components of the system can enable quantification of climate adaptation effectiveness as well as residual risk (remaining risk after adaptation) to ecosystems and fisheries. Some of our ongoing CCS research focuses on specific species groups (i.e. forage species) but with better integration of model components. Models in development include: a



Fig. 9. (a) Maps of indices for historical (1995–2018) harvesting and processor engagement in the albacore fishery, for medium sized vessels (45–60 ft). Each unit of the index represents one standard deviation, meaning that a value of 8 represents extreme engagement compared to the mean level of engagement. The histograms under each map show the distribution of index values for the most engaged 100 communities (harvesting index) or all 67 engaged communities (processor index). (b) Projected proportional change in port-level albacore landings for medium vessels for two future periods compared to the 1995–2018 historical period, for the three ROMS-ESMs.

California Current Atlantis spatial ecosystem model (Kaplan et al., 2017) with dynamic, climate-responsive spatial distribution of key species or functional groups informed by SDMs; an ensemble of sardine operating models for more tactical MSEs; improved bioeconomic modeling for projections of impacts on the CPS fishery; and a spatial population dynamics model for sardine including environment-driven movement.

We are also exploring how changes in forage fish will propagate through the CCS food web, based not only on ecosystem modeling (and the planned Atlantis modeling incorporating species distributions) but also trait-based modeling. Trait-based ecology focuses on the role of organism traits in mediating organism abundance and distribution, and how these traits can be used to predict unobserved responses (beyond using only abiotic variables). Body size, habitat associations, and life history traits are commonly used when investigating climate change effects (Green et al., 2022), but interactions between species and their impacts to species productivity and distributions are an often-overlooked biotic filtering step. By assessing predator diets based on traits of forage species that synthesize trophic interactions, we can generate future predictions of predator distributions given trophic filtering processes such as prey switching. These approaches will be applied to research on albacore, which exhibit dynamic prey switching behavior in response to varying forage availability, and we aim to evaluate whether trait-based models, and measurements of prey energy content, can improve projections of albacore distributions and potentially abundance.

The coupled ocean-biogeochemical model projections developed as part of Future Seas have provided a foundation for the wide array of ecological and socio-economic analyses described herein, as well as in a number of other CCS research projects. Nonetheless, there are a number of avenues for enhancements and improvements to these projections. Modifications to the model domain could improve the utility of the projections; e.g. a larger domain would enable better coverage of species ranges and connectivity, and higher spatial resolution could improve representation of fine scale processes and bathymetry, particularly over the continental shelf. Inclusion of river inputs in future versions of the model projections would improve their fidelity especially in the northern part of the domain, and would widen the scope of applications to include species that are dependent on estuarine or freshwater sources at the land-sea interface. Lastly, as is typical of regional ocean simulations, the size of our projection ensemble (three ESMs, one RCP) was limited by computational resources. Including more ESMs in the projection ensemble, as well as forcing from multiple scenarios, would allow for improved characterization of the range of potential ocean futures. This is a key goal for regional climate projections intended for marine resource applications, but one that often competes with efforts to enhance resolution (Drenkard et al., 2021). Our experiences in Future Seas lead us to prioritize greater domain size and more ensemble members over further increases to ocean model resolution.

The CCS is a well-studied system, and there are numerous studies using climate projections to investigate the physical, ecological and socio-economic future of the CCS (e.g. Ainsworth et al., 2011; Woodworth-Jefcoats et al., 2013; Cheung et al., 2015; Marshall et al., 2017; Morley et al., 2018; Haltuch et al., 2019; Howard et al., 2020). There are also ongoing efforts beyond our own to develop projections for the CCS, for nearby regions, and globally, which are relevant to our results and project structure and to the development of integrated projection analyses in general. Important foci for these modeling efforts include: evaluation of relative climate risk to coastal communities and fisheries (Payne et al., 2021); integrated, iterative, and multi-model frameworks (e.g. the Alaska Climate Integrated Modeling project; Hollowed et al., 2020), which can develop projections while robustly estimating uncertainty (Reum et al., 2020; Whitehouse et al., 2021); developing projections for species complexes in the CCS, including groundfish, and highly migratory species (e.g. the Fisheries and Climate Toolkit Project, https://fisheriesclimatetoolkit.sdsu.edu); and development of accessible and standardized ensembles of climate-fishery models, to encourage

consistent and comparable projections and uncertainty (i.e. Fisheries and Marine Ecosystem Model Intercomparison Project; Tittensor et al., 2018, Tittensor et al., 2021). Marine ecosystems are complex, and there are many issues and processes to explore in projection studies (and beyond what is currently considered in Future Seas), including food web changes and the balance of nutrients, vitamins (e.g. thiamine; Sutherland et al., 2018; Mantua et al., 2021), and predator–prey dynamics (Ohlberger et al., 2019). Multi-model frameworks are likely the best way to evaluate these complex systems, by balancing narrower focused models for tactical projection analysis alongside generalized and ecosystem model for strategic projection analysis.

7. Recommendations

7.1. Consider the physical, ecological, and socio-economic context

The general methodological approach of Future Seas - linking climate models to oceanographic, ecological, and socioeconomic models - can be employed in a wide range of applications and regions. However, in each case the details of the modeling framework should be tailored to the specific context; outlining key elements of that context, including stakeholder priorities, is thus an important early step in such a project design. For Future Seas, the physical, ecological, and socio-economic context is outlined in Section 3, and many of the elements described here will be applicable to other projects. Examples include the spatiotemporal scales of dominant atmospheric and oceanographic processes, prominence of anthropogenic signals relative to natural variability, population dynamics of focal species (e.g., boom-bust vs more stable), bottom-up vs top-down control of ecological change, regulatory environment and management concerns (e.g., harvest guidelines, spatial management, species interactions or human-wildlife conflict), and socio-economic considerations (e.g., cultural and economic value, large vs small-scale fisheries). Of course, other projects will likely have similar considerations as well as different or additional ones. In any case, an effort to thoroughly understand the context of a project at its outset will help to ensure that methods, analyses, and products are ultimately responsive to, and able to address, stakeholder concerns.

7.2. Develop a modeling and analysis framework guided by the context and constraints

With a thorough understanding of the context in which the project is being carried out, an appropriate analytical framework can then be developed. We have found that a thoughtful and intentional approach to this process can be aided by some guiding questions: Which models should be used? How will environmental information be incorporated? Which management modeling framework is best suited to addressing stakeholder concerns? What are the relevant metrics of change produced by the models? We provide details of these considerations for Future Seas in Section 4 as a case study. In brief, some examples include (i) matching oceanographic projections to population dynamics of focal species (e.g., projecting continuous ocean change, as opposed to just past and future slices, is appropriate for species such as sardine whose populations exhibit dramatic low-frequency variability), (ii) matching ecological model type to the characteristics of focal species (e.g., we included population dynamics models for sardine, which spawn in the CCS, but focused on distribution models for swordfish, which spawn outside the CCS), and (iii) matching management strategies to stakeholder concerns (e.g., alternative harvest guidelines for sardine, alternative spatiotemporal closures for swordfish).

7.3. Fit and project ecological models with appropriate environmental information

When making ecological projections based on environmental data, it is necessary to reconcile historical data used to develop ecological models with the model output used for long-term projection. Future climate projections are typically derived from ESMs that are global in scope, and include coupled atmosphere, land and ocean components that are freely evolving in response to greenhouse gases and aerosols. In these simulations, the statistics (i.e. mean and variability) of quantities of interest under historical greenhouse gas levels (e.g. the 1980–2020 period of a 1980–2100 projection) should match observations, but projection output from specific years in this historical period would not match the historical observations (Stock et al., 2011; Drenkard et al., 2021; Table 1). For example, the historical portion of a climate model simulation will have El Niño events, but their timing will not match those in the historical record. While this fact is not a shortcoming of the models, it does severely limit the use of the historical period of the projection for ecological model fitting.

As a result, historical ocean state estimates used to fit ecological models often come from either in situ observations, a model hindcast, or a reanalysis (Table 1). Each of these data sources has advantages and disadvantages in the context of developing ecological projections. reanalyses combine the strengths of observations and hindcasts to give ocean state estimates that have complete spatiotemporal coverage and are more accurate than those from a hindcast. However, they are typically physics-only, are much more time and computationally intensive to create, and perform best for regions, times, and variables that are well constrained by observations. The inclusion of biogeochemical dynamics is much more common in ocean hindcasts, but the absence of data assimilation is, however, likely to degrade the ocean state estimate to some degree.

A recommended workflow for model fitting-projection would be: (1) fit ecological models based on historical environmental data (observations, hindcast, or reanalysis, depending on their availability and suitability); (2) project the fitted model, using ocean model output forced by ESMs, including historical and future periods (e.g., 1980–2100); (3) check that the mean/variability in the projections of the ecological/ economic quantity of interest over the historical period is comparable to the one obtained when predicting with realistic historical data; and (4) reference future changes to the historical period of the ESM-driven projection. If step 3 shows that future changes are referenced to a historical model simulation that is biased relative to reality, it is more appropriate to interpret changes (e.g. fishing catch) in relative rather than absolute terms.

7.4. Capture uncertainty and use ensembles

When models are used to predict the real world, it is important that outputs include estimates of uncertainty and error in predictions. Uncertainty and error will always exist and arise from imperfect understanding, representation, or observation of the system being modeled. There are numerous schemes for classifying and distinguishing uncertainty and error, and these will often be idiosyncratic to the modeling approach and study. Some effort has been made to unify the schemes for climate and ecological models, by using four main classifications: structural (or model) uncertainty, parametric uncertainty, scenario uncertainty, and internal (and sometimes initialization) uncertainty (Cheung et al., 2016; Payne et al., 2016). Fisheries science, and often for MSEs, typically uses some combination of six classifications: process, observation, parameter, structural (or model), estimation (or assessment), and implementation uncertainty (Francis and Shotton, 1997; Punt et al., 2016).

In Table 3 we specify these main sources of variation - attempting to align the two classification schemes above - and how they were, or could be, represented in our project. Accounting for every source of uncertainty in any projection is probably not feasible due to computational and statistical limitations, and attempting to do so could increase the potential to overestimate joint uncertainty, conceal important signals, and challenge the ability to provide climate and management advice. Instead, a reasonable goal is to identify the most important

biases

Table 3

Types of uncertainty and how they were, or could be, incorporated in our projections. These broad classifications are often divided into more specific classifications. For example, SDM uncertainty can be estimated separately from ESM uncertainty (Morleyet al., 2020) even though both are types of model uncertainty. ¹Francis and Shotton (1997), ²Hill et al. (2007), ³Payne et al. (2016).

Type of uncertainty	Domain relevance	Description	Representation in Future Seas	Example methods to estimate/incorporate
Scenario	Physical, socio- economic	Uncertainty arising from forcing processes, typically future emissions and aerosols arising from future socio-economic patterns; could also include 'implementation uncertainty' ¹ , if scenarios were used to model variable uptake of management actions	Climate: not evaluated (used single RCP) Other: alternate spatial closure responses evaluated in swordfish models	Evaluate multiple RCPs or SSPs (shared socio-economic pathways); develop fishing industry and management scenarios
Model / Structural	Physical, ecological, socio-economic	Uncertainty arising from differences in model design, configuration, or structure	Climate: three ESMs usedOther: multiple SDMs used (swordfish), multiple model types used (sardine) and multiple configurations (sardine pop. dynamics model)	Multiple models or 'ensembles'; model weights can also be explored ²
Parametric / Parameter	Physical, ecological, socio-economic	Arising from variation in model parameters; compared to model uncertainty (which is about evaluating various plausible structures), parameter uncertainty focuses on how well a model explains data (e.g., uncertainty due to extrapolation of SDMs)	Climate: implicit in use of three ESMs Other: multiple iterations of the swordfish MSE, varying key parameters and sampling posterior distributions	Run models multiple times with resampling of parameter estimates ² ; report mean and spread of model outputs; model extrapolation can also be measured as a surrogate of statistical measures of parameter (and prediction) uncertainty
Internal variability / process uncertainty	Physical, ecological, socio-economic	Natural variations that contribute to 'random' variability in projections (e.g. stochasticity in population dynamics); can also include 'initialization uncertainty' ³ , which is both an observation and parameter problem; internal variability could be considered to be estimated by model residuals, although they maay be unrealistically large in poorly specified models; true internal variability is considered 'irreducible uncertainty' ¹	Climate: implicit in three ESMs with different internal variabilityOther: in the albacore projection, future fishing locations were randomly assigned, constrained by historical distances offshore and effort within an SST- chlorophyll envelope; in the swordfish MSE, catches were sampled randomly from an empirical distribution, representing irreducible fishing catch success (i.e. they caught the mean catch only in the long run)	Multiple iterations of climate projections from the same scenario and model; allow for and estimate with error terms (e.g. recruitment deviations in stock assessment models); run models multiple times ²
Observation	Physical, ecological, socio-economic	Uncertainty associated with imperfect measurement and sampling of the system; more often explored in ecological models	Evaluated in the swordfish MSE by random variation in the simulated observer program, and by ensuring simulated observations were as accurate of the simulated truth as real observations are in reality	Incorporate through iteration; specify observer error in estimation model of MSE; estimate using hierarchical modeling where appropriate; reduce by increasing amount and coverage of observation; evaluate and communicate observation



Fig. 10. Relative uncertainty in center of gravity (COG) predictions for the Pacific sardine population in the CCS, partitioned across earth systems models, model configuration (for the population dynamics model), and model type. COGs are smoothed over 10 years. Analysis was completed using the *dominanceAnalysis* function in the 'dominanceanalysis' R package (Navarrete and Soares 2020), with model configuration nested within model type.

uncertainties, propagate them through the projections, and ensure the interpretation of results acknowledges the variation and uncertainty that remains unmodeled. Our most thorough evaluation of uncertainty was for sardine, specifically the center of gravity (COG) projections (Fig. 5), for which we can partition the relative contributions of: uncertainty associated with the ESMs; uncertainty associated with ecological model type (SDM, IBM, population dynamics); and uncertainty associated with multiple configurations of a single model type (the population dynamics model). We found that the relative contribution of uncertainty associated with ecological model type increased as the projection horizon increased, while the relative ESM uncertainty decreased (Fig. 10). Our results for Pacific sardine agree in part with those of Brodie et al. (2022) and Reum et al. (2020), who found that ecological model uncertainty increased over time, although Reum et al. (2020) also found a similar increase in ESM uncertainty for some species groups.

Based on analyses to date and subsequent exploration of uncertainty (Table 3, Fig. 10), we report the following findings and recommendations:

- Our projections would benefit from additional emission scenarios (or Shared Socio-economic Pathways; Section 7.5); however, these may not be the dominant source of uncertainty, especially for shorter projection horizons (Frölicher et al., 2016), because our suite of models for the CCS under RCP8.5 captured much of the range in projected physical and biogeochemical change exhibited in lower RCPs (Pozo Buil et al., 2021).
- Our project, and likely others, would benefit from developing and integrating scenarios for processes other than just emissions, such as socio-economic trajectories that might influence factors such as seafood price or fishing effort and efficiency (see Section 7.5).
- Our projections would benefit from additional ESMs. While the three we selected bound the range of potential futures well, we are unable to quantify the likelihood of different trajectories within that range. Ensembles of models are useful, although care should be taken when determining model inclusion, as not all models are equally reasonable (Overland et al., 2011; Muhling et al., 2018). Deciding whether to report mean results from an ensemble, or results from each member of an ensemble, is an important consideration; model weighting (Hill et al., 2007) may be appropriate when presenting ensemble mean results, but can be challenging to implement objectively.
- Ecological model uncertainty is a large source of uncertainty, potentially equal to or greater than climate model uncertainty in long projections (Fig. 10, Thuiller et al., 2019; Morley et al., 2020), highlighting the value of developing multiple ecological model configurations and parameterizations (Smith et al., 2021b, Koenigstein et al. in review), and potentially different model types that are diverse in fundamental structure (Fig. 5).
- The uncertainty of correlative SDM projections will be underestimated by assuming stationarity of habitat associations (e.g. ignoring acclimation or adaptation). Although this uncertainty can be identified with careful cross-validation (Muhling et al., 2020), how this uncertainty can be propagated into projection results is unclear. Building multiple SDMs may (inadvertently) help to account for this uncertainty. A future research focus could be a better representation of non-stationarity in SDMs, perhaps with hybrid modeling approaches that incorporate dynamic processes in correlative models (e.g. Bush et al., 2016).
- Extrapolation is a key aspect of projection, and is a key issue for correlative SDMs, especially given the emergence rate of climate novelty (Smith et al., 2022; Brodie et al., 2022; Fig. 4). Poor extrapolation is a parameterization problem, but it is unclear how this uncertainty can be represented in model results. Mapping where and when extrapolation occurs (Zurell et al., 2012) can identify where models may be less accurate, even if the magnitude of that

inaccuracy is uncertain. Quantifying biological rates (e.g. migration capacity, stage-specific survival, consumption and growth, or predation rates) in process-based models could help to validate or constrain extrapolations.

- When future uncertainty is high, constructing and interpreting future scenarios can be simplified by projecting change in only some variables while keeping others constant. For example, in our SDM-based sardine landings simulation only the environmentally-driven sardine distributions are projected, and our interpretation is restricted to the impacts of environmental change on a representative but largely independent fishing industry. An alternative to this could be carefully controlled scenarios, especially in terms of socio-economics, which we discuss in the next section.
- More and better data should always be a priority, and benefits modeling in a number of ways. including: reducing (and quantifying) uncertainty in data such as the locations of fishing catches; better constraining and evaluation of ocean models; and improving empirical associations between species and their environment.

7.5. Consider scenarios and socio-economic futures

Socio-economic variables are some of the most difficult to characterize and project (Fulton et al., 2011). In Future Seas to date, we typically assumed constant: fishing effort and behavior, port capacity, gear efficiency, fish price, etc; and used this realistic background to isolate environmentally-driven change. This was also the case in the engagement analyses of the albacore fishery, because it was unreasonable to attempt to project engagement (given projected albacore changes) without information on how vessel ownership, port infrastructure, or albacore ex-vessel value might change or respond to these albacore changes. In short, our projection results are fundamentally based on a static socio-economic future, when adaptation and innovation will likely be key for reducing potential environmental impacts, and the future ecosystem state will depend on feedbacks from a changed human system.

A valuable alternative to attempting to model these extremely complex processes is to use socio-economic scenarios. A key development of the research community is 'shared socio-economic pathways' (Riahi et al., 2017) - which have been applied to aquatic systems in similar integrated climate and fisheries projects (Pinnegar et al., 2020) and the adapted 'oceanic system pathways' (Maury et al., 2017). SSPs are one component of scenarios integrating future changes in climate (including emissions) and society, and encompass five core narratives and associated quantitative descriptions encompassing future change in human demographics, development, economy, institutions, technology, and environment (O'Neill et al., 2017). A similar effort was made for the U.S. West Coast as part of the PFMC's Climate and Communities Initiative (PFMC, 2020): experts derived four narratives predicting potential changes to 2040, distinguishable primarily along two axes representing changes in climate and species abundance. These global, ocean, and West Coast scenarios currently serve as signposts for potential physical and socio-economic futures to help bound projections, but progress is needed before these scenarios can be linked directly to climate, ecological, and social projections, namely by defining realistic (and hopefully generalizable) quantitative variables and values associated with each scenario (Maury et al., 2017; Boschetti et al., 2020).

An alternative to strictly quantitative modeling, but also offering strategic advice, is the Climate Vulnerability Assessment framework, which is designed to systematically identify potential climate change impacts on fisheries social-ecological systems (Hare et al., 2016). This framework has been demonstrated prospectively for the U.S. West Coast albacore fishery, given projected climate changes (Dudley et al., 2021). That study did not develop socio-economic scenarios, but rather used characteristics of current ecological and social subsystems to identify areas of the socio-economic system most sensitive to future change, where the greatest uncertainties lie, and (in the case of albacore)

anticipate potential interjurisdictional conflicts.

8. Conclusion

Our downscaled climate projections show that, without strong curbing of emissions, the CCS will undergo significant change this century, including 2–4 °C warming of SST and an almost ubiquitous shift to novel conditions (Pozo Buil et al., 2021; Smith et al., 2022). However, there is a considerable difference in patterns of change among ESMs, which highlights the value of model ensembles. Our case studies highlight that issues of species redistribution will be paramount, which will impact spatial closures, quota allocations, fleet and port accessibility, and community engagement. Biomass will also be influenced, but we are only beginning to understand, and represent in models, the relationship between biomass and climate change. It seems likely that engagement of U.S. West Coast communities with specific species and fisheries will change in the coming decades, and flexibility in both fishing grounds and catch portfolios will be important processes for community resilience. The value of such long-term projection information is related to understanding the potential extent of change and impacts which necessarily determines the resilience of our natural and human systems. and to decadal-scale decision making such as long-term investments in infrastructure and industry. The NOAA Fisheries Climate Science Strategy was developed to aid the preparation and response to climaterelated impacts on U.S. living marine resources and resource-dependent communities (Busch et al., 2016), and some key objectives of this strategy are identifying future states of ecosystems and communities, and building science infrastructure. We hope that Future Seas has contributed, and will continue to contribute, to meeting both these objectives by developing and sharing long-term projections of the CCS and important fisheries. Coordination across integrated climate analyses and consistent delivery of climate-informed fisheries advice, like that from Future Seas, is the foundation of effective adaptation and climate ready fisheries (Bell et al., 2020; IPCC, 2022). Our estimates of long-term changes and impacts complement more tactical scales (i.e. real-time monitoring and modeling, and near-term forecasts) to contribute to a comprehensive view of the future CCS. This supporting role of long-term projection helps provide as much information as possible to stakeholders and decision makers, including helping to identify the more vulnerable and more resilient - species, fisheries, management strategies, and communities.

Declaration of Competing Interest

The authors declare the following financial interests/personal relationships which may be considered as potential competing interests: EC and RRR serve as editors for Progress in Oceanography.

Data availability

Data availability is reported in Appendix C. Some remains confidential and cannot be made public.

Acknowledgements

Funding was provided by NOAA's Coastal and Ocean Climate Application COCA Program, United States (NA17OAR4310268); NOAA's Climate and Fisheries Adaptation CAFA Program, United States (NA20OAR4310507); NOAA's Integrated Ecosystem Assessment Program, United States, and the National Aeronautics and Space Administration, United States (NASA; 80NSSC19K0187). For assistance with data access, we are grateful to the Pacific Fisheries Information Network (particularly J. Suter; https://pacfin.psmfc.org), the California Department of Fish and Wildlife (particularly E. Hellmers), the Oregon Department of Fish and Wildlife (particularly J. Ainsworth), the Washington Department of Fish and Wildlife (particularly C. Niles), and J. Rakitan at UCSC.

Author contributions

MGJ, MA, EC, RRR, AHC, SJB, JF, ELH, SS, DT, BM, AH and CAS initiated the Future Seas project, and MGJ led project administration. All authors contributed to project development and various aspects of the analyses. Ocean projections were led by MPB and MGJ, sardine analyses by JAS, BM, JF and SK, swordfish analyses by JAS, SB, SS, NLO and HW, and albacore analyses by BM, DT, TF and AMC. JAS and MGJ led the writing, and all authors contributed to the original draft and its revision.

Appendix A. Supplementary material

Supplementary data to this article can be found online at https://doi.org/10.1016/j.pocean.2023.102973.

References

- Ainsworth, C.H., Samhouri, J.F., Busch, D.S., Cheung, W.W., Dunne, J., Okey, T.A., 2011. Potential impacts of climate change on Northeast Pacific marine foodwebs and fisheries. ICES Journal of Marine Science 68 (6), 1217–1229.
- Aminpour, P., Gray, S.A., Jetter, A.J., Introne, J.E., Singer, A., Arlinghaus, R., 2020. Wisdom of stakeholder crowds in complex social–ecological systems. Nat. Sustainability 3 (3), 191–199.
- Barange, M., Coetzee, J., Takasuka, A., Hill, K., Gutierrez, M., Oozeki, Y., Agostini, V., 2009. Habitat expansion and contraction in anchovy and sardine populations. Prog. Oceanogr. 83, 251–260.
- Baumgartner, T., Soutar, A., Ferreira-Bartrina, V., 1992. Reconstruction of the history of Pacific sardine and northern anchovy populations over the past two millennia from sediments of the Santa Barbara Basin, California. CalCOFI Report 33, 24–40.
- Bell, R.J., Odell, J., Kirchner, G., Lomonico, S., 2020. Actions to promote and achieve climate-ready fisheries: summary of current practice. Mar. Coast. Fish. 12 (3), 166–190.
- Bernard, H.R., 2017. Research Methods in Anthropology: Qualitative and Quantitative Approaches, fourth edition. Rowman & Littlefield, London.
- Blenckner, T., Llope, M., Möllmann, C., Voss, R., Quaas, M.F., Casini, M., Lindegren, M., Folke, C., Chr. Stenseth, N., 2015. Climate and fishing steer ecosystem regeneration to uncertain economic futures. Proc. R. Soc. B Biol. Sci. 282 (1803), 20142809.
- Block, B.A., Jonsen, I.D., Jorgensen, S.J., Winship, A.J., Shaffer, S.A., Bograd, S.J., Costa, D.P., 2011. Tracking apex marine predator movements in a dynamic ocean. Nature 475, 86–90.
- Bograd, S.J., Jacox, M.G., Hazen, E.L., Lovecchio, E., Montes, I., Pozo Buil, M., Shannon, L.J., Sydeman, W.J., Rykaczewski, R.R., 2023. Climate change impacts on eastern boundary upwelling systems. Ann. Rev. Mar. Sci. 15, 303–328.
- Boschetti, F., Bulman, C.M., Hobday, A.J., Fulton, E.A., Contardo, S., Lozano-Montes, H., Ingrid van Putten, E., 2020. Sectoral futures are conditional on choices of global and national scenarios–Australian marine examples. Front. Mar. Sci. 935.
- Brady, R.X., Alexander, M.A., Lovenduski, N.S., Rykaczewski, R.R., 2017. Emergent anthropogenic trends in California Current upwelling. Geophysical Research Letters 44 (10), 5044–5052.
- Brodie, S., Jacox, M.G., Bograd, S.J., Welch, H., Dewar, H., Scales, K.L., Hazen, E.L., 2018. Integrating dynamic subsurface habitat metrics into species distribution models. Front. Mar. Sci. 5.
- Brodie, S., Abrahms, B., Bograd, S.J., Carroll, G., Hazen, E.L., Muhling, B.A., Jacox, M.G., 2021. Exploring timescales of predictability in species distributions. Ecography 44, 832–844.
- Brodie, S., Smith, J.A., Muhling, B.A., Kaplan, I.C., Barnett, L.A.K., Carroll, G., Zabel, R., 2022. Recommendations for quantifying and reducing uncertainty in projections of species distributions. Glob. Chang. Biol. 28, 6586–6601.
- Busch, D.S., Griffis, R., Link, J., Abrams, K., Baker, J., Brainard, R.E., Merrick, R., 2016. Climate science strategy of the US national marine fisheries service. Mar. Policy 74, 58–67.
- Bush, A., Mokany, K., Catullo, R., Hoffmann, A., Kellermann, V., Sgro, C., Ferrier, S., 2016. Incorporating evolutionary adaptation in species distribution modelling reduces projected vulnerability to climate change. Ecol. Lett. 19 (12), 1468–1478.
- Cavole, L.M., Demko, A.M., Diner, R.E., Giddings, A., Koester, I., Pagniello, C.M., Franks, P.J., 2016. Biological impacts of the 2013–2015 warm-water anomaly in the Northeast Pacific: winners, losers, and the future. Oceanography 29 (2), 273–285.
- Checkley, D.M., Barth, J.A., 2009. Patterns and processes in the California Current System. Prog. Oceanogr. 83, 49–64.
- Cheung, W.W., Brodeur, R.D., Okey, T.A., Pauly, D., 2015. Projecting future changes in distributions of pelagic fish species of Northeast Pacific shelf seas. Progress in Oceanography 130, 19–31.

Cheung, W.W.L., Frölicher, T.L., Asch, R.G., Jones, M.C., Pinsky, M.L., Reygondeau, G., Watson, J.R., 2016. Building confidence in projections of the responses of living marine resources to climate change. ICES J. Mar. Sci. 73, 1283–1296.

Cheung, W.W.L., 2018. The future of fishes and fisheries in the changing oceans. J. Fish Biol. 92 (3), 790–803.

Colburn et al., 2017. Community Participation in U.S. Catch Share Programs. U.S. Dept. of Commerce, NOAA. NOAA Technical Memorandum NMFS-F/SPO-179, 136 p.

Cooley, S., Schoeman, D., Bopp, L., Boyd, P., Donner, S., Ghebrehiwet, D.Y., Ito, S.-I., Kiessling, W., Martinetto, P., Ojea, E., Racault, M.-F., Rost, B., Skern-Mauritzen M., 2022. Ocean and coastal ecosystems and their services. In: Pörtner,H.-O., Roberts, D. C., Tignor, M., Poloczanska, E.S., Mintenbeck, K., Alegría, A., Craig, M., Langsdorf, S., Löschke, S., Möller, V., Okem, A., Rama, B. (Eds.), Climate Change 2022: Impacts, Adaptation, and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press (In Press).

Cordero-Quirós, N., Jacox, M.G., Pozo-Buil, M., Bograd, S.J. (in review) Future changes in eddy kinetic energy in the California Current System from dynamically downscaled climate projections. Geophys. Res. Lett.

- Cramer, L.A., Flathers, C., Caracciolo, D., Russell, S.M., Conway, F., 2018. Graying of the fleet: Perceived impacts on coastal resilience and local policy. Mar. Policy 96, 27–35. Deyle, E.R., Fogarty, M., Hsieh, C.-H., Kaufman, L., MacCall, A.D., Munch, S.B.,
- Sugihara, G., 2013. Predicting climate effects on Pacific sardine. Proc. Natl. Acad. Sci. 110, 6430–6435.
- Ding, Q., Chen, X., Hilborn, R., Chen, Y., 2017. Vulnerability to impacts of climate change on marine fisheries and food security. Mar. Policy 83 (May), 55–61.

Doney, S.C., Ruckelshaus, M., Emmett Duffy, J., Barry, J.P., Chan, F., English, C.A., Talley, L.D., 2012. Climate change impacts on marine ecosystems. Ann. Rev. Mar. Sci. 4, 11–37.

Drenkard, E.J., Stock, C., Ross, A.C., Dixon, K.W., Adcroft, A., Alexander, M., Wang, M., 2021. Next-generation regional ocean projections for living marine resource management in a changing climate. ICES J. Mar. Sci. 78, 1969–1987.

du Pontavice, H., Gascuel, D., Reygondeau, G., Maureaud, A., Cheung, W.W.L., 2020. Climate change undermines the global functioning of marine food webs. Glob. Chang. Biol. 26, 1306–1318.

- Dudley, P.N., Rogers, T.L., Morales, M.M., Stoltz, A.D., Sheridan, C.J., Beulke, A.K., Pomeroy, C., Carr, M.H., 2021. A more comprehensive climate vulnerability assessment framework for fisheries social-ecological systems. Frontiers in Marine Science 8, 674.
- Dunn, D.C., Maxwell, S.M., Boustany, A.M., Halpin, P.N., 2016. Dynamic ocean management increases the efficiency and efficacy of fisheries management. Proc. Natl. Acad. Sci. 113, 668–673.
- Essington, T.E., Moriarty, P.E., Froehlich, H.E., Hodgson, E.E., Koehn, L.E., Oken, K.L., Stawitz, C.C., 2015. Fishing amplifies forage fish population collapses. Proc. Natl. Acad. Sci. 112, 6648–6652.
- Fiechter, J., Rose, K.A., Curchitser, E.N., Hedstrom, K.S., 2015. The role of environmental controls in determining sardine and anchovy population cycles in the California Current: Analysis of an end-to-end model. Prog. Oceanogr. 138, 381–398.
- Fiechter, J., Pozo Buil, M., Jacox, M.G., Alexander, M.A., Rose, K.A., 2021. Projected Shifts in 21st Century Sardine Distribution and Catch in the California Current. Front. Mar. Sci. 8, 685241.
- Fisher, M.C., Moore, S.K., Jardine, S.L., Watson, J.R., Samhouri, J.F., 2021. Climate shock effects and mediation in fisheries. PNAS 118 (2), 1–8.
- Francis, R., Shotton, R., 1997. "Risk" in fisheries management: a review. Can. J. Fish. Aquat. Sci. 54, 1699–1715.
- Frawley, T.H., Muhling, B.A., Brodie, S., Fisher, M.C., Tommasi, D., Le Fol, G., Jacox, M. G., 2021. Changes to the structure and function of an albacore fishery reveal shifting social-ecological realities for Pacific Northwest fishermen. Fish Fish. 22, 280–297.
- Free, C.M., Thorson, J.T., Pinsky, M.L., Oken, K.L., Wiedenmann, J., Jensen, O.P., 2019. Impacts of historical warming on marine fisheries production. Science 363 (6430), 979–983
- Frölicher, T.L., Rodgers, K.B., Stock, C.A., Cheung, W.W., 2016. Sources of uncertainties in 21st century projections of potential ocean ecosystem stressors. Global Biogeochem. Cycles 30 (8), 1224–1243.
- Fuller, E.C., Samhouri, J.F., Stoll, J.S., Levin, S.A., Watson, J.R., 2017. Characterizing fisheries connectivity in marine social–ecological systems. ICES J. Mar. Sci. 74 (8), 2087–2096.

Fulton, E.A., Smith, A.D.M., Smith, D.C., Van Putten, I.E., 2011. Human behaviour: The key source of uncertainty in fisheries management. Fish Fish. 12 (1), 2–17.

Fulton, E.A., 2010. Approaches to end-to-end ecosystem models. J. Mar. Syst. 81, 171–183.

Green, S.J., Brookson, C.B., Hardy, N.A., Crowder, L.B., 2022. Trait-based approaches to global change ecology: moving from description to prediction. Proceedings of the Royal Society B 289 (1971), 20220071.

Haltuch, M.A., A'mar, Z.T., Bond, N.A., Valero, J.L., 2019. Assessing the effects of climate change on US West Coast sablefish productivity and on the performance of alternative management strategies. ICES J. Mar. Sci. 76 (6), 1524–1542.

Hare, J.A., Morrison, W.E., Nelson, M.W., Stachura, M.M., Teeters, E.J., Griffis, R.B., et al., 2016. A Vulnerability assessment of fish and invertebrates to climate change on the Northeast U.S. continental shelf. PLoS One 11 (2), e0146756.

Harvey, C.J., Kelble, C.R., Schwing, F.B., 2017. Implementing "the IEA": using integrated ecosystem assessment frameworks, programs, and applications in support of operationalizing ecosystem-based management. ICES Journal of Marine Science 74 (1), 398–405.

Harvey, C., Garfield, N., Williams, G., Tolimieri, N., Andrews, K., Barnas, K., Zwolinski, J., 2020. Ecosystem Status Report of the California Current for 2019-20: A Summary of Ecosystem Indicators Compiled by the California Current Integrated Ecosystem Assessment Team (CCIEA).

Hazen, E.L., Jorgensen, S., Rykaczewski, R.R., Bograd, S.J., Foley, D.G., Jonsen, I.D., Block, B.A., 2013. Predicted habitat shifts of Pacific top predators in a changing climate. Nat. Clim. Chang. 3 (3), 234–238.

Progress in Oceanography 211 (2023) 102973

Hill, S.L., Watters, G.M., Punt, A.E., McAllister, M.K., Quéré, C.L., Turner, J., 2007. Model uncertainty in the ecosystem approach to fisheries. Fish Fish. 8, 315–336.

- Himes-Cornell, A., Kasperski, S., 2016. Using socio-economic and Fisheries Involvement Indices to Understand Alaska Fishing Community Well-Being. Coast. Manag. 44, 36–70.
- Holland, D.S., Leonard, J., 2020. Is a delay a disaster? economic impacts of the delay of the california dungeness crab fishery due to a harmful algal bloom. Harmful Algae 98, 101904.

Hollowed, A.B., Bond, N.A., Wilderbuer, T.K., Stockhausen, W.T., A'mar, Z.T., Beamish, R.J., Schirripa, M.J., 2009. A framework for modelling fish and shellfish responses to future climate change. ICES J. Mar. Sci. 66, 1584–1594.

Hollowed, A.B., Holsman, K.K., Haynie, A.C., Hermann, A.J., Punt, A.E., Aydin, K., Wilderbuer, T.K., 2020. Integrated modeling to evaluate climate change impacts on coupled social-ecological systems in Alaska. Front. Mar. Sci., 6.

Holsman, K., Samhouri, J., Cook, G., Hazen, E., Olsen, E., Dillard, M., Andrews, K., 2017. An ecosystem-based approach to marine risk assessment. Ecosyst. Health Sustainability 3, e01256.

Horne, P.J., Kaplan, I.C., Marshall, K.N., Levin, P.S., Harvey, C.J., Hermann, A.J. and Fulton, E.A., 2010. Design and parameterization of a spatially explicit ecosystem model of the central California Current. NOAA technical memorandum NMFS-NWFSC-104.

Howard, E.M., Penn, J.L., Frenzel, H., Seibel, B.A., Bianchi, D., Renault, L., Kessouri, F., Sutula, M.A., McWilliams, J.C., Deutsch, C., 2020. Climate-driven aerobic habitat loss in the California Current System. Science advances 6 (20), eaay3188.

- IATTC, 2017. Tunas, billfishes, and other pelagic species in the eastern Pacific Ocean in 2016. Inter-American Tropical Tuna Commission, 92nd meeting, Mexico 2017. Document IATTC-92-04a.
- IPCC, 2021a. Annex VII: Glossary. In: Matthews, J.B.R., Möller, V., van Diemen, R., Fuglestvedt, J.S., Masson-Delmotte, V., Méndez, C., Semenov, S., Reisinger, A. (Eds.), Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change [Masson-Delmotte, V., P. Zhai, A. Pirani, S.L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M.I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J.B.R. Matthews, T. K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, and B. Zhou (eds.)]. Cambridge University Press Cambridge, United Kingdom and New York, NY, USA, pp. 2215–2256, doi: 10.1017/9781009157896.022.
- IPCC, 2021b. Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change [Masson-Delmotte, V., P. Zhai, A. Pirani, S.L. Connors, C. Péan, S. Berger, N. Caud, Y. Chen, L. Goldfarb, M.I. Gomis, M. Huang, K. Leitzell, E. Lonnoy, J.B.R. Matthews, T.K. Maycock, T. Waterfield, O. Yelekçi, R. Yu, and B. Zhou (Eds.)]. Cambridge University Press (in press).
- IPCC, 2022. Climate Change 2022: Impacts, Adaptation, and Vulnerability. Contribution of Working Group II to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change [H.-O. Pörtner, D.C. Roberts, M. Tignor, E.S. Poloczanska, K. Mintenbeck, A. Alegría, M. Craig, S. Langsdorf, S. Löschke, V. Möller, A. Okem, B. Rama (Eds.)]. Cambridge University Press (in press).
- ISC Albacore Working Group, 2021. Report of the North Pacific Albacore Tuna Management Strategy Evaluation. 21st Meeting of the International Scientific Committee for Tuna and Tuna-Like Species in the North Pacific Ocean. Electronic meeting. https://meetings.wcpfc.int/node/12593.

ISC Billfish Working Group, 2018. Stock assessment for swordfish (Xiphias gladius) in the Western and Central North Pacific Ocean through 2016. Scientific Committee Fourteenth Regular Session, Western and Central Pacific Fisheries Commission. Busan, Republic of Korea. http://isc.fra.go.jp/pdf/ISC18/ISC_18_ANNEX_16_Stock_ Assessment_of_WCNPO_Swordfish_through_2016_FINAL.pdf.

Jacobson, L.D., MacCall, A.D., 1995. Stock-recruitment models for Pacific sardine (Sardinops sagax). Can. J. Fish. Aquat. Sci. 52, 566–577.

- Jepson M, Colburn LL. Development of social indicators of fishing community vulnerability and resilience in the US Southeast and Northeast regions. 2013. NOAA Technical Memorandum NMFS-F/SPO-129.
- Kaplan, I.C., Koehn, L.E., Hodgson, E.E., Marshall, K.N., Essington, T.E., 2017. Modeling food web effects of low sardine and anchovy abundance in the California Current. Ecol. Model. 359, 1–24.

Kaplan, I.C., Gaichas, S.K., Stawitz, C.C., Lynch, P.D., Marshall, K.N., Deroba, J.J., Link, J., 2021. Management strategy evaluation: allowing the light on the hill to illuminate more than one species. Front. Mar. Sci. 8, 688.

Kasperski, S., Holland, D.S., 2013. Income diversification and risk for fishermen. Proc. Natl. Acad. Sci. 110, 2076–2081.

Kasperski, S., DePiper, G.S., Haynie, A.C., Blake, S., Colburn, L.L., Freitag, A., Weijerman, M., 2021. Assessing the State of Coupled Social-Ecological Modeling in Support of Ecosystem Based Fisheries Management in the United States. Front. Mar. Sci. 8.

King, J.R., Agostini, V.N., Harvey, C.J., McFarlane, G.A., Foreman, M.G.G., Overland, J. E., Di Lorenzo, E., Bond, N.A., Aydin, K.Y., 2011. Climate forcing and the California Current ecosystem. ICES J. Mar. Sci. 68 (6), 1199–1216.

Kishi, M.J., et al., 2007. NEMURO - a lower trophic level model for the North Pacific marine ecosystem. Ecol. Model. 202 (1–2), 12–25.

- Koehn, L.E., Essington, T.E., Marshall, K.N., Kaplan, I.C., Sydeman, W.J., Szoboszlai, A.I., Thayer, J.A., 2016. Developing a high taxonomic resolution food web model to assess the functional role of forage fish in the California Current ecosystem. Ecol. Model. 335, 87–100.
- Koenigstein, S., Jacox, M.G., Pozo Buil, M., Fiechter, J., Muhling, B.A., Brodie, S., Kuriyama, P., Auth, T.D., Hazen, E.L., Bograd, S.J., Tommasi, D., 2022. Population projections of Pacific sardine driven by ocean warming and changing food

J.A. Smith et al.

availability in the California Current. ICES J. Mar. Sci. https://doi.org/10.1093/ icesjms/fsac191.

- Koslow, J.A., Davison, P., Lara-Lopez, A., Ohman, M.D., 2014. Epipelagic and mesopelagic fishes in the southern California Current System: Ecological interactions and oceanographic influences on their abundance. J. Mar. Syst. 138, 20–28.
- Kuriyama, P.T., Zwolinski, J.P., Hill, K.T., Crone, P.R. Assessment of the Pacific sardine resource in 2020 for U.S. management in 2020–2021. Pacific Fishery Management Council, Portland, OR. Available from https://www.pcouncil.org/coastal-pelagicspecies/stock-assessmentand-fshery-evaluation-safe-documents/.
- Lam, V.W.Y., Cheung, W.W.L., Reygondeau, G., Rashid Sumaila, U., 2016. Projected change in global fisheries revenues under climate change. Sci. Rep. 6, 6–13.
- Lindegren, M., Checkley, D.M., Rouyer, T., MacCall, A.D., Stenseth, N.C., 2013. Climate, fishing, and fluctuations of sardine and anchovy in the California Current. Proc. Natl. Acad. Sci. 110, 13672–13677.
- Lotze, H.K., Tittensor, D.P., Bryndum-Buchholz, A., Eddy, T.D., Cheung, W.W., Galbraith, E.D., Worm, B., 2019. Global ensemble projections reveal trophic amplification of ocean biomass declines with climate change. Proc. Natl. Acad. Sci. 116 (26), 12907–12912.
- Mackinson, S., Wilson, D.C., Galiay, P., Deas, B., 2011. Engaging stakeholders in fisheries and marine research. Mar. Policy 35 (1), 18–24.
- Madigan, D.J., Carlisle, A.B., Dewar, H., Snodgrass, O.E., Litvin, S.Y., Micheli, F., Block, B.A., 2012. Stable Isotope Analysis Challenges Wasp-Waist Food Web Assumptions in an Upwelling Pelagic Ecosystem. Sci. Rep. 2, 654.
- Mahony, C.R., Cannon, A.J., Wang, T., Aitken, S.N., 2017. A closer look at novel climates: new methods and insights at continental to landscape scales. Glob. Chang. Biol. 23 (9), 3934–3955.
- Mantua, N., Johnson, R., Field, J., Lindley, S., Williams, T., Todgham, A., Ruiz-Cooley, I., 2021. Mechanisms, Impacts, and Mitigation for Thiamine Deficiency and Early Life Stage Mortality in California's Central Valley Chinook Salmon. North Pacific Anadromous Fish Commission, 92–93.
- Marshall, K.N., Kaplan, I.C., Hodgson, E.E., Hermann, A., Busch, D.S., McElhany, P., Fulton, E.A., 2017. Risks of ocean acidification in the California Current food web and fisheries: ecosystem model projections. Glob. Chang. Biol. 23, 1525–1539.
- Mason, J.G., Hazen, E.L., Bograd, S.J., Dewar, H., Crowder, L.B., 2019. Community-level effects of spatial management in the California drift gillnet Fishery. Fish. Res. 214, 175–182.
- Maury, O., Campling, L., Arrizabalaga, H., Aumont, O., Bopp, L., Merino, G., Van Ruijven, B.J., 2017. From shared socio-economic pathways (SSPs) to oceanic system pathways (OSPs): Building policy-relevant scenarios for global oceanic ecosystems and fisheries. Glob. Environ. Chang. 45, 203–216.
- McCabe, R.M., Hickey, B.M., Kudela, R.M., Lefebvre, K.A., Adams, N.G., Bill, B.D., Trainer, V.L., 2016. An unprecedented coastwide toxic algal bloom linked to anomalous ocean conditions. Geophys. Res. Lett. 43, 10,366-310,376.
- McDaniel, J., Piner, K., Lee, H.-H., Hill, K., 2016. Evidence that the Migration of the Northern Subpopulation of Pacific Sardine (Sardinops sagax) off the West Coast of the United States Is Age-Based. PLoS One 11, e0166780.
- McEvoy, A.F., 1986. The fisherman's problem: ecology and law in the California fisheries, 1850–1980. Cambridge University Press.
- McFarlane, G., Smith, P., Baumgartner, T. & Hunter, J., 2002. Climate variability and Pacific sardine populations and fisheries. American Fisheries Society Symposium, pp. 195–214. American Fisheries Society.
- Morley, J.W., Selden, R.L., Latour, R.J., Frölicher, T.L., Seagraves, R.J., Pinsky, M.L., 2018. Projecting shifts in thermal habitat for 686 species on the North American continental shelf. PLoS One 13, e0196127.
- Morley, J.W., Frölicher, T.L., Pinsky, M.L., 2020. Characterizing uncertainty in climate impact projections: a case study with seven marine species on the North American continental shelf. Ices. J. Mar. Sci.
- Muhling, B.A., Gaitán, C.F., Stock, C.A., Saba, V.S., Tommasi, D., Dixon, K.W., 2018. Potential salinity and temperature futures for the Chesapeake Bay using a statistical downscaling spatial disaggregation framework. Estuar. Coasts 41 (2), 349–372.
- Muhling, B., Brodie, S., Snodgrass, O., Tommasi, D., Dewar, H., Childers, J., Snyder, S., 2019. Dynamic habitat use of albacore and their primary prey species in the California Current system. CalCOFI Report 60, 79–93.
- Muhling, B.A., Brodie, S., Smith, J.A., Tommasi, D., Gaitan, C.F., Hazen, E.L., Brodeur, R. D., 2020. Predictability of species distributions deteriorates under novel environmental conditions in the California Current System. Front. Mar. Sci. 7.
- Navarrete, C.B., Soares, F.C., 2020. dominanceanalysis: Dominance Analysis. R package version 2.0.0. https://CRAN.R-project.org/package=dominanceanalysis.
- Neveu, E., Moore, A.M., Edwards, C.A., Fiechter, J., Drake, P., Crawford, W.J., Nuss, E., 2016. An historical analysis of the California Current circulation using ROMS 4D-Var: System configuration and diagnostics. Ocean Model. 99, 133–151.
- Nieto, K., Xu, Y., Teo, S.L., McClatchie, S., Holmes, J., 2017. How important are coastal fronts to albacore tuna (Thunnus alalunga) habitat in the Northeast Pacific Ocean? Prog. Oceanogr. 150, 62–71.
- Ohlberger, J., Schindler, D.E., Ward, E.J., Walsworth, T.E., Essington, T.E., 2019. Resurgence of an apex marine predator and the decline in prey body size. Proc. Natl. Acad. Sci. 116 (52), 26682–26689.
- Ong, J.J.L., Walter, J.A., Jensen, O.P., Pinsky, M.L., 2021. Global hotspots of coherent marine fishery catches. Ecol. Appl. 31, e02321.
- Overland, J.E., Wang, M., Bond, N.A., Walsh, J.E., Kattsov, V.M., Chapman, W.L., 2011. Considerations in the selection of global climate models for regional climate projections: the Arctic as a case study. J. Clim. 24 (6), 1583–1597.
- O'Neill, B.C., Kriegler, E., Ebi, K.L., Kemp-Benedict, E., Riahi, K., Rothman, D.S., van Ruijven, B.J., van Vuuren, D.P., Birkmann, J., Kok, K., Levy, M., 2017. The roads ahead: Narratives for shared socioeconomic pathways describing world futures in the 21st century. Global Cnvironmental Change 42, 169–180.

- Payne, M.R., Barange, M., Cheung, W.W.L., MacKenzie, B.R., Batchelder, H.P., Cormon, X., Paula, J.R., 2016. Uncertainties in projecting climate-change impacts in marine ecosystems. ICES J. Mar. Sci. 73, 1272–1282.
- Payne, M.R., Kudahl, M., Engelhard, G.H., Peck, M.A., Pinnegar, J.K., 2021. Climate risk to European fisheries and coastal communities. Proc. Natl. Acad. Sci. U.S.A. 118(40), 1–10.
- Pecl, G.T., Araújo, M.B., Bell, J.D., Blanchard, J., Bonebrake, T.C., Chen, I.-C., Williams, S.E., 2017. Biodiversity redistribution under climate change: impacts on ecosystems and human well-being. Science, 355, eaai9214.
- PFMC, 2020. 2040: Scenarios for West Coast Fisheries. Climate and Communities Initiative, Pacific Fishery Management Council's Climate and Communities Core Team. https://www.pcouncil.org/documents/2020/11/scenarios-for-west-coast-fish eries-climate-and-communities-initiative.pdf/.
- PFMC, 2022. Pacific Coast Fishery Ecosystem Plan for the U.S. Portion of the California Current Large Marine Ecosystem (Revised and Updated). Pacific Fishery Management Council, Portland, Oregon.
- Pinnegar, J.K., Hamon, K.G., Kreiss, C., Tabeau, A., Rybicki, S., Papathanasopoulou, E., Engelhard, G.H., Eddy, T.D., Peck, M.A., 2020. Future socio-political scenarios for aquatic resources in Europe: a common framework based on shared-socioeconomicpathways (SSPs). Front. Mar. Sci. 7, 1096.
- Poloczanska, E.S., Brown, C.J., Sydeman, W.J., Kiessling, W., Schoeman, D.S., Moore, P. J., Richardson, A.J., 2013. Global imprint of climate change on marine life. Nat. Clim. Chang. 3, 919–925.
- Polovina, J.J., Howell, E., Kobayashi, D.R., Seki, M.P., 2001. The transition zone chlorophyll front, a dynamic global feature defining migration and forage habitat for marine resources. Prog. Oceanogr. 49 (1–4), 469–483.
- Pons, M., Watson, J. T., Ovando, D., Andraka, S., Brodie, S., Domingo, A., Hilborn, R., 2022. Trade-offs between bycatch and target catches in static versus dynamic fishery closures. Proc. Natl. Acad. Sci. 119(4).
- Pozo Buil, M., Jacox, M.G., Fiechter, J., Alexander, M.A., Bograd, S.J., Curchitser, E.N., Stock, C.A., 2021. A dynamically downscaled ensemble of future projections for the California current system. Front. Mar. Sci. 8.
- Punt, A.E., Butterworth, D.S., de Moor, C.L., De Oliveira, J.A., Haddon, M., 2016. Management strategy evaluation: best practices. Fish Fish. 17 (2), 303–334.
- Reglero, P., Tittensor, D.P., Álvarez-Berastegui, D., Aparicio-González, A., Worm, B., 2014. Worldwide distributions of tuna larvae: revisiting hypotheses on environmental requirements for spawning habitats. Mar. Ecol. Prog. Ser. 501, 207–224.
- Reum, J.C., Blanchard, J.L., Holsman, K.K., Aydin, K., Hollowed, A.B., Hermann, A.J., Cheng, W., Faig, A., Haynie, A.C., Punt, A.E., 2020. Ensemble projections of future climate change impacts on the Eastern Bering Sea food web using a multispecies size spectrum model. Frontiers in Marine Science 7, 124.
- Riahi, K., Rao, S., Krey, V., Cho, C., Chirkov, V., Fischer, G., Rafaj, P., 2011. RCP 8.5—A scenario of comparatively high greenhouse gas emissions. Clim. Change 109 (1), 33–57.
- Riahi, K., Van Vuuren, D.P., Kriegler, E., Edmonds, J., O'neill, B.C., Fujimori, S., Bauer, N., Calvin, K., Dellink, R., Fricko, O., Lutz, W., 2017. The shared socioeconomic pathways and their energy, land use, and greenhouse gas emissions implications: an overview. Global Environmental Change 42, 153–168.
- Richerson, K., Holland, D.S., 2017. Quantifying and predicting responses to a US West Coast salmon fishery closure. ICES J. Mar. Sci. 74, 2364–2378.
- Rose, K.A., Fiechter, J., Curchitser, E.N., Hedstrom, K., Bernal, M., Creekmore, S., Agostini, V., 2015. Demonstration of a fully-coupled end-to-end model for small pelagic fish using sardine and anchovy in the California Current. Prog. Oceanogr. 138, 348–380.
- Russell, S.M., Oostenburg, M.V., Vizek, A., 2018. Adapting to catch shares: Perspectives of West Coast groundfish trawl participants. Coast. Manag. 46 (6), 603–620.
- Ryabinin, V., Barbière, J., Haugan, P., Kullenberg, G., Smith, N., McLean, C., Rigaud, J., 2019. The UN Decade of Ocean Science for Sustainable Development. Front. Mar. Sci. 6.
- Rykaczewski, R.R., Dunne, J.P., 2010. Enhanced nutrient supply to the California Current Ecosystem with global warming and increased stratification in an earth system model. Geophys. Res. Lett. 37.
- Rykaczewski, R.R., Dunne, J.P., Sydeman, W.J., García-Reyes, M., Black, B.A., Bograd, S. J., 2015. Poleward displacement of coastal upwelling-favorable winds in the ocean's eastern boundary currents through the 21st century. Geophys. Res. Lett. 42, 6424-6431.
- Samhouri, J.F., Ramanujam, E., Bizzarro, J.J., Carter, H., Sayce, K., Shen, S., 2019. An ecosystem-based risk assessment for California fisheries co-developed by scientists, managers, and stakeholders. Biol. Conserv. 231, 103–121.
- Santora, J.A., Mantua, N.J., Schroeder, I.D., Field, J.C., Hazen, E.L., Bograd, S.J., Forney, K.A., 2020. Habitat compression and ecosystem shifts as potential links between marine heatwave and record whale entanglements. Nat. Commun. 11, 536.
- Savoca, M.S., Brodie, S., Welch, H., Hoover, A., Benaka, L.R., Bograd, S.J., Hazen, E.L., 2020. Comprehensive bycatch assessment in US fisheries for prioritizing management. Nat. Sustainability 3 (6), 472–480.
- Scales, K.L., Hazen, E.L., Jacox, M.G., Castruccio, F., Maxwell, S.M., Lewison, R.L., Bograd, S.J., 2018. Fisheries bycatch risk to marine megafauna is intensified in Lagrangian coherent structures. Proc. Natl. Acad. Sci. 115, 7362–7367.
- Sepulveda, C.A., Aalbers, S.A., Heberer, C., Kohin, S., Dewar, H., 2018. Movements and behaviors of swordfish Xiphias gladius in the United States Pacific Leatherback Conservation Area. Fish. Oceanogr. 27, 381–394.
- Sepulveda, C.A., Wang, M., Aalbers, S.A., Alvarado-Bremer, J.R., 2020. Insights into the horizontal movements, migration patterns, and stock affiliation of California swordfish. Fisheries Oceanography 29 (2), 152–168.

J.A. Smith et al.

Siple, M.C., Essington, T.E., Barnett, L.A.K., Scheuerell, M.D., 2020. Limited evidence for sardine and anchovy asynchrony: re-examining an old story. Proc. R. Soc. B Biol. Sci. 287, 20192781.

- Smith, J.A., Tommasi, D., Sweeney, J., Brodie, S., Welch, H., Hazen, E.L., Jacox, M.G., 2020. Lost opportunity: Quantifying the dynamic economic impact of time-area fishery closures. J. Appl. Ecol. 57, 502–513.
- Smith, J.A., Muhling, B., Sweeney, J., Tommasi, D., Pozo Buil, M., Fiechter, J., Jacox, M. G., 2021a. The potential impact of a shifting Pacific sardine distribution on U.S. West Coast landings. Fish. Oceanogr. 30, 437–454.
- Smith, J.A., Tommasi, D., Welch, H., Hazen, E.L., Sweeney, J., Brodie, S., Jacox, M.G., 2021b. Comparing dynamic and static time-area closures for bycatch mitigation: a management strategy evaluation of a swordfish fishery. Front. Mar. Sci. 8.
- Smith, J.A., Pozo Buil, M., Fiechter, J., Tommasi, D., Jacox, M.G., 2022. Projected novelty in the climate envelope of the California Current at multiple spatial-temporal scales. PLOS Climate 1 (4), e0000022.

Smith, P.E., 2005. A history of proposals for subpopulation structure in the Pacific sardine (Sardinops sagax) population off western North America. Cal. Cooper. Ocean. Fish. Invest. Rep. 46, 75–82.

Stock, C.A., Alexander, M.A., Bond, N.A., Brander, K.M., Cheung, W.W.L., Curchitser, E. N., Werner, F.E., 2011. On the use of IPCC-class models to assess the impact of climate on Living Marine Resources. Prog. Oceanogr. 88, 1–27.

Sutherland, W.J., Butchart, S.H., Connor, B., Culshaw, C., Dicks, L.V., Dinsdale, J., Gleave, R.A., 2018. A 2018 horizon scan of emerging issues for global conservation and biological diversity. Trends Ecol. Evol. 33 (1), 47–58.

Thompson, A.R., Harvey, C.J., Sydeman, W.J., Barceló, C., Bograd, S.J., Brodeur, R.D., Williams, G.D., 2019. Indicators of pelagic forage community shifts in the California Current Large Marine Ecosystem, 1998–2016. Ecol. Ind. 105, 215–228. Thuiller, W., Guéguen, M., Renaud, J., Karger, D.N., Zimmermann, N.E., 2019.

 Indiret, W., Ougach, W., Ichardy, G., Rarger, D.N., Zhimrenham, W.J., 2019.
 Uncertainty in ensembles of global biodiversity scenarios. Nat. Commun. 10, 1446.
 Tittensor, D.P., Eddy, T.D., Lotze, H.K., Galbraith, E.D., Cheung, W., Barange, M.,
 Walker, N.D., 2018. A protocol for the intercomparison of marine fishery and ecosystem models: Fish-MIP v1.0. Geosci. Model Dev. 11, 1421–1442.

- Tittensor, D.P., Novaglio, C., Harrison, C.S., Heneghan, R.F., Barrier, N., Bianchi, D., Blanchard, J.L., 2021. Next-generation ensemble projections reveal higher climate risks for marine ecosystems. Nat. Clim. Chang. 11, 973–981.
- Ueber, E, MacCall, A, 1992. The rise and fall of the California sardine empire. In: Glantz, MH (Ed.), Climate Variability, Climate Change and Fisheries. Cambridge Univ. Press, Cambridge, UK, pp. 31–48.
- Urbisci, L.C., Stohs, S.M., Piner, K.R., 2016. From sunrise to sunset in the California Drift Gillnet Fishery: an examination of the effects of time and area closures on the catch and catch rates of pelagic species. Mar. Fish. Rev. 78, 1–11.

- Veneziani, M., Edwards, C.A., Doyle, J.D., Foley, D., 2009. A central California coastal ocean modeling study: 1. Forward model and the influence of realistic versus climatological forcing, J. Geophys. Res. Oceans 114.
- Weber, E.D., Auth, T.D., Baumann-Pickering, S., Baumgartner, T.R., Bjorkstedt, E.P., Bograd, S.J., Zeman, S.M., 2021. State of the California Current 2019–2020: Back to the Future With Marine Heatwaves? Front. Mar. Sci., 8.
- Weiskopf, S.R., Harmáčková, Z.V., Johnson, C.G., Londoño-Murcia, M.C., Miller, B.W., Myers, B.J., Rosa, I.M., 2022. Increasing the uptake of ecological model results in policy decisions to improve biodiversity outcomes. Environ. Model. Softw. 149, 105318.
- Welch, H., Hazen, E.L., Briscoe, D.K., Bograd, S.J., Jacox, M.G., Eguchi, T., Bailey, H., 2019. Environmental indicators to reduce loggerhead turtle bycatch offshore of Southern California. Ecol. Ind. 98, 657–664.
- Wells, B.K., Schroeder, I.D., Bograd, S.J., Hazen, E.L., Jacox, M.G., Leising, A., Peterson, W.T., 2017. State Of The California Current 2016–17: Still anything but" normal" in the North. Cal. Cooper. Ocean. Fish. Invest. Rep. 58, 1–55.
- Whitehouse, G.A., Aydin, K.Y., Hollowed, A.B., Holsman, K.K., Cheng, W., Faig, A., Haynie, A.C., Hermann, A.J., Kearney, K.A., Punt, A.E., Essington, T.E., 2021.
 Bottom–Up Impacts of Forecasted Climate Change on the Eastern Bering Sea Food Web. Frontiers in Marine Science 8, 624301.
- Williams, J.W., Jackson, S.T., Kutzbach, J.E., 2007. Projected distributions of novel and disappearing climates by 2100 AD. Proceedings of the National Academy of Sciences 104 (14), 5738–5742.
- Woodworth-Jefcoats, P.A., Polovina, J.J., Dunne, J.P., Blanchard, J.L., 2013. Ecosystem size structure response to 21st century climate projection: large fish abundance decreases in the central North Pacific and increases in the California Current. Glob. Chang. Biol. 19, 724–733.
- Xiu, P., Chai, F., Curchitser, E.N., Castruccio, F.S., 2018. Future changes in coastal upwelling ecosystems with global warming: The case of the California Current System. Sci. Rep. 8, 2866.
- Zainuddin, M., Saitoh, K., Saitoh, S.I., 2008. Albacore (Thunnus alalunga) fishing ground in relation to oceanographic conditions in the western North Pacific Ocean using remotely sensed satellite data. Fish. Oceanogr. 17 (2), 61–73.
- Zurell, D., Elith, J., Schröder, B., 2012. Predicting to new environments: tools for visualizing model behaviour and impacts on mapped distributions. Divers. Distrib. 18 (6), 628–634.
- Zwolinski, J.P., Emmett, R.L., Demer, D.A., 2011. Predicting habitat to optimize sampling of Pacific sardine (Sardinops sagax). ICES J. Mar. Sci. 68, 867–879.